University of Arkansas, Fayetteville ScholarWorks@UARK

Theses and Dissertations

5-2012

Perceived Impact of Quality in a 21st Century Community Learning Center Out-of-School Time Program: A Case Study

Patricia Ann Wright University of Arkansas, Fayetteville

Follow this and additional works at: http://scholarworks.uark.edu/etd Part of the Educational Assessment, Evaluation, and Research Commons, Educational Leadership Commons, and the Elementary Education and Teaching Commons

Recommended Citation

Wright, Patricia Ann, "Perceived Impact of Quality in a 21st Century Community Learning Center Out-of-School Time Program: A Case Study" (2012). *Theses and Dissertations*. 369. http://scholarworks.uark.edu/etd/369

This Dissertation is brought to you for free and open access by ScholarWorks@UARK. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of ScholarWorks@UARK. For more information, please contact scholar@uark.edu, ccmiddle@uark.edu.

PERCEIVED IMPACT OF QUALITY IN A 21ST CENTURY COMMUNITY LEARNING CENTER OUT-OF-SCHOOL TIME PROGRAM: A CASE STUDY

PERCEIVED IMPACT OF QUALITY IN A 21st CENTURY COMMUNITY LEARNING CENTER OUT-OF-SCHOOL TIME PROGRAM: A CASE STUDY

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Education in Educational Leadership

By

Patricia Thomas Wright Henderson State University Bachelor of Science in Education, 1982 Henderson State University Master of Science in Education, 1986 Henderson State University Education Specialist, 2005

> May 2012 University of Arkansas

ABSTRACT

Concern for the safety, education and well-being of children during out-of-school time is helping to change the landscape and priorities of families, educators and policy-makers. Changes in family structure and society have presented the need for quality out-of-school time programs. The changing family structure caused by both parents working outside the home, the advent of the single-parent household, the necessity of federally mandated standardized testing for student achievement, certain criminal activities and the expanding population of children have contributed to the question of how to protect children and enrich their lives during out-of-school time hours. The purpose of this intrinsic case study is to describe the key elements of quality in out-of-school time programs for school age youth as perceived by the participants and to determine the impact of the quality standards in out-of-school time programs as perceived by administrators, teachers, and students. The program participants are middle school students based on academic performance enrolled in a 21st Century Community Learning Center out-of-school time program in Southeast Arkansas.

Participants included a diverse population from various socioeconomic backgrounds that were enrolled in the program because of basic or below scores on the benchmark examination. The study is designed around the premise out-of-school time programs built around key elements of quality reinforce outcomes of student achievement, personal enrichment and out-of-school time programs matter. Interview content regarding student achievement was found to be the richest category with related themes of grades and homework. Interview data revealed shared themes regarding the program for the participants' category.

The study provides insight into quality elements of out-of-school time programs, specifically, student achievement through gains made in homework and grades. The case study

identified elements of quality that support student achievement outcomes for school-based programs including positive program perceptions, sustainable funding, and building strong interpersonal relationships. The findings suggest key elements of quality were identified and may contribute to positive outcomes for students. Program sustainability was a major concern for staff and the future of the OST program. This study contributes to the data needed to identify out-of-school time quality elements across program types and geographical locations. This dissertation is approved for recommendation to the Graduate Council.

Dissertation Director:

Carleton R. Holt, Ed.D.

Dissertation Committee:

Ed Bengtson, Ph.D.

Betsy Orr, Ed.D.

DISSERTATION DUPLICATION RELEASE

I hereby authorize the University of Arkansas Libraries to duplicate this dissertation when needed for research and/or scholarship.

Agreed

Patricia A. Wright

Refused

Patricia A. Wright

ACKNOWLEDGEMENTS

With sincere gratitude and appreciation for the "village" that provided the foundation and support to complete this journey. Through the lessons learned in the village, I realized that all things are indeed possible, if you keep the faith.

I want to express my sincere appreciation to my chair, Dr. Carleton Holt, for his leadership, guidance and patience. Special thanks to my current and former committee members Dr. Orr, Dr. Bengtson and Dr. Deggs for your willingness to serve on my committee and provide much needed assistance and support. I would also like to acknowledge and say thank you to my loving Pastors, Kenneth and Claire Rhodes for your prayers, encouragement and editorial support.

A special acknowledgement and thanks to the southeast school district administrators, teachers, and students for allowing me to complete my study in your district. Thank you for your contribution to the study and your commitment and dedication to your students. I am also very appreciative to my school board and the administration for allowing the time and providing the support to complete this project.

Last, but certainly not least, my family and friends that are too numerous to name for being such a wonderful, caring, supportive family and lifetime friends. I am truly humble and grateful to those that have paved the way for me and my desire is to do the same for others.

"To God be the glory for the things He has done."

DEDICATION

This work is dedicated in memory of my parents and grandparents:

To my husband Vern, for your unconditional love and the dedication you have shown me. To my children and grandchildren, thank you for your patience, kindness and the love that you have shown during my journey. To my parents and grandparents, the lessons and values that were taught and instilled in me at an early age provided the foundation for aligning my life by putting God first, family second, and third, becoming a lifelong learner to be of service to others. To my family, friends and neighbors for being a part of the village that provided continuous encouragement, love and support.

TABLE OF CONTENTS

ABSTRACT

LIST OF TABLES

LIST OF FIGURES

| Chapter One: Introduction | 1 |
|--|----|
| Background of the Study | 2 |
| Statement of the Problem | 3 |
| Purpose of the Study | 6 |
| Significance of the Study | 6 |
| Research Questions | 7 |
| Research design | 7 |
| Research setting | 7 |
| Theoretical Framework | |
| Limitation of the Study | |
| Definition of Terms | |
| Conclusion | |
| Chapter Two: Review of the Literature | 13 |
| Background | 14 |
| Methods | 16 |
| Search Strategy | 17 |
| Historical Perspective | |
| No Child Left Behind Act of 2001 | |
| Out-of-school time Programs | |
| Out-of-school time Program Evaluations | |

| Impact of OST Programs | |
|--|----|
| Theoretical Framework | |
| Descriptive elements of program quality | |
| States' quality standards | |
| Significance | |
| Summary of Review of Literature | |
| Conclusion | |
| Chapter Three: Methodology | 41 |
| Research Questions | 43 |
| Research design | 43 |
| Researcher's role | |
| Data Management | 53 |
| Data collection | 54 |
| Data analysis | |
| Trustworthiness and credibility | |
| Prolonged engagement | |
| Persistent engagement | |
| Triangulation | |
| Member checks | |
| Audit trail | |
| Summary | |
| Chapter Four: Data Presentation and Analysis | 60 |
| Introduction | |
| Description of Participants | 61 |

| Data Management | 62 |
|---|----|
| Research question one | 64 |
| Elements of quality | 65 |
| Description of Themes and Categories | 66 |
| Student participants | 66 |
| Student achievement | 72 |
| Program sustainability and outcomes | 76 |
| Relationships | 78 |
| Research question two | 81 |
| Summary of Data | 83 |
| Chapter Five: Conclusions and Recommendations | 84 |
| Introduction | 84 |
| Summary | 84 |
| Research question one | 85 |
| Research question two | 86 |
| Interpretation of Data | 87 |
| Program Recommendations | 89 |
| Recommendations to the Field | 90 |
| Recommendations for Future Research | 91 |
| Conclusion | 92 |
| References | 95 |
| Appendix | |

LIST OF TABLES

| Table | Page |
|---|------|
| Table 1 States with Quality Standards | |
| Table 2 Sample Interview Questions | |
| Table 3 Field Note Guide | |
| Table 4 2008-2009 Middle School Enrollment | |
| Table 5 Annual School Performance Report 2009 | |
| Table 6 Distribution of Categories & Themes | 57 |
| Table 7 Description of Participants | |
| Table 8 Audit Trail Notations | |
| Table 9 Elements of Quality | |

LIST OF FIGURES

| Figure | Page |
|---|------|
| Figure 1. Quality elements for out-school time programs | 9 |
| Figure 2. African-American performance indicators in science, literacy, and | |
| mathematics | 51 |
| Figure 3. Comparison: African-American vs. economically disadvantaged middle school | |
| students | 52 |

Chapter One

Introduction

Out-of-school time (OST) programs protect children and youth from victimization and delinquency during what law enforcement officials' term "danger zone" hours between 3 and 7 p.m. (Office of Juvenile Justice and Delinquency Prevention [OJJDP], 2006). Programs offer students rewarding, challenging, age and developmentally appropriate activities in a safe, structured, supervised environment. Students receive remediation in core academic areas where they may be struggling, assistance with homework and a chance to choose from a variety of enrichment programs that support growth and development of youth. OST programs support student achievement and address risks associated with dropping out of school. Some of these OST programs are possible through grant funding, fees, and private contributions.

The largest federal funding source is the 21st Century Community Learning Centers made possible by the No Child Left Behind Act of 2001 (NCLB, 2002). Federal dollars were appropriated to states for disbursement through a grant application process. The 21st Century Community Learning Centers (21st CCLC) out-of-school time program provides extended learning opportunities before and after-school for school-age children in public schools, community, and faith-based organizations. Programs offer opportunities for students to receive homework assistance, academic remediation, a chance for students to choose which enrichment activities interests to engage in, and an opportunity to participate in recreational and community activities. The programs are helping to address the change in family structure that often create the problem of unsupervised children and youth during out-of-school time by keeping children safe and in an environment of learning and enrichment until parents return home from work (U.S. Department of Education, 2002). The desired outcome of an out-of-school time program is to improve the status of schools and academic performance of students. According to the Arkansas Department of Education reports, the school district selected for this case study was listed on the "School Improvement" list. Schools in need of improvement failed to meet adequate yearly progress (AYP) goals for two consecutive years therefore listed as in need of improvement-Year 1 and must offer public school choice according to No Child Left Behind Act of 2001 (NCLB, 2002). Continual improvements over two consecutive years allowed the middle school to be removed from the state's list of problem schools because of assistance received from Yale University as a 21st Century Charter Member and as a 21st CCLC grantee. Student achievement improved at the middle school as reflected by test scores, homework completion and attendance. However, progress for economically disadvantaged and African-American students was slow (Beebe, 2008).

Background of the Study

Former Arkansas Governor Mike Huckabee initiated a Governor's Summit on Extra Learning Opportunities to address the need to expand, support, and improve the quality of outof-school time (National Governors Association, 2006). Huckabee opened the dialogue for programs to receive statewide support allowing Arkansas Governor Mike Beebe to continue the work two years later by establishing his Task Force on Best Practices for After-School and Summer Programs. The Governor's Task Force released a report entitled, Enriching Arkansas Children's Lives Through High-Quality Out-of-School Activities (Beebe, 2008) that identified key elements for quality programs utilized by other states and national organizations. The Arkansas Out-of-School Network (AOSN) aligned their quality standards to the quality framework issued by the Governor's Task Force.

The Demand Study for After-School Programs conducted in 2008 by students from the University of Arkansas Clinton School of Public Service indicated half of 3,700 students surveyed in Arkansas were unsupervised after-school on a regular basis. The study concluded "regardless of gender, race, or grade level, a large portion of students are unsupervised after-school," particularly older students (Guzzardi, Little, & Mitchell, 2008, p. 13). Key recommendations summarized from the study included:

- 1. Efforts for program recruitment should focus on students that are unsupervised afterschool, particularly middle school students.
- 2. Additional after-school programs may be needed in various parts of the state to address the needs of children and families.
- 3. The state should also focus on funding sources for after-school programs to ensure that programs are accessible and affordable for families.

Society's investment toward the success of a child who becomes a taxpayer and college graduate versus the cost to society for one child in the criminal justice system reflects the benefits of out-of-school time programs. Out-of-school time programs benefit children and youth by increasing safety during out-of-school hours, reducing risk-taking behavior, and providing a safe, structured and supervised learning environment.

Statement of the Problem

Changes in American society since the industrial revolution have presented the need for quality out-of-school time care for children and youth. The changing family structure caused by

both parents working outside the home, the advent of the single-parent household, the necessity of federally mandated standardized testing for student achievement, criminal activity and the expanding population of children, have contributed to the problem of how to protect children and enrich their lives during out-of-school time hours.

Traditionally, the composition of the family consisted of a two-parent household, one parent serving as the caretaker, and the other parent working outside of the home. In the modern family, both parents work with sixty-nine percent (69%) of all married couples with children ages 6-17 working outside the home and seventy-nine point five percent (79.5%) of single parents employed outside the home (OJJDP, 2006). The change in family economic structure from one breadwinner to two has impacted childcare in the United States. In 2004, twenty-five percent (25%) of U.S. children took care of themselves after-school while only eight percent of Arkansas children were in self-care during after-school hours. In 2009, national and Arkansas statistics were the same; twenty-six percent (26%) of children are in self-care after-school (Afterschool Alliance, 2004).

The growing public interest regarding how and where children spend their time during out-of-school time because of the increased number of working parent households has sparked the need for quality out-of-school time programs. The definition of out-of-school time referred to both traditional after-school programs operating in the afternoon hours and programs with "more comprehensive efforts that respond to the needs of children, youth, and parents during evenings, weekends, summers, and holidays by offering activities that help youth grow, learn, and develop" (American Youth Policy Forum, 2006, p. 5). A shift from academic remediation and the safety aspects of after-school programs has occurred toward out-of-school time

programming with comprehensive efforts to serve young people and produce positive outcomes for program participants. OST programs are offered by various organizations including schools, community and faith-based organizations, libraries, museums, municipalities, and volunteer groups seeking to serve youth.

However, the difference in family structure creates hardships for poor families regardless of the type of organization that offers OST programs. Statistics depict the disparity in income required to cover basic needs. Forty-five point nine percent (45.9 %) of poor single parent families reported hardships associated with unmet basic needs compared to thirty-eight point six percent (38.6 %) of poor two parent families. In families reporting no hardships, twenty-three point three percent (23.3%) were single-parent families compared to forty-one point two percent (41.2 %) of two parent families (Beverly, 2001). Female headed single-parent households are more at risk for higher levels of stress; have less social support overall including less contact with friends and family and less involvement in church and social groups. In fact, single mothers experience depression twice as often as married mothers (Carney & Boyle, 2003).

Also, living in households without both biological parents augment risk factors for children and youth. Middle school age youth living in a household without both biological parents have four times the risk of developing an affective disorder (Cuff, Keown, Addy & Garrison, 2005). Single mothers raising children alone increase the risk for teenage pregnancy, marrying before obtaining a high school diploma and marrying someone without a high school diploma (Teachman, 2004).

Purpose of the Study

Now, more than ever, an increasing body of research focusing on the safety, education and well being of children during out-of-school time is helping to restructure the national policy agenda for parents, educators, and policymakers throughout the United States. The purpose of this case study is to describe the key elements of quality in out-of-school time programs for school age youth as perceived by the participants. The program participants are middle school students based on academic performance enrolled in a 21st Century Community Learning Center afterschool program in Southeast Arkansas. The faculty includes a Site Coordinator, certified teachers, paraprofessionals and peer tutors. A 21st Century Community Learning Center OST program in southeast Arkansas is the target of this case study.

Out-of-school time programs are measured by student attendance and academic achievement. Students who become bored, dissatisfied with academic improvement, or disengaged in program activities sometimes leave the program and may never return. Students stay engaged in program activities that best meet their academic and personal development needs. Quality OST programs utilize student interest surveys that provide students a choice of enrichment activities and a voice in the program structure.

Significance of the Study

Benefits of quality out-of-school time programs can be explained by looking at the cost of society's investment toward the success of a child that is able to contribute to society as a taxpayer and college graduate versus the cost to society for one child in the criminal justice system. A high-risk student that benefits from attending a community program that encourages scholastic achievement becomes a college graduate with a meaningful career and is able to benefit society by making a 1 million dollar tax contribution during his lifetime (Trostil, 2007). A high-risk fourteen year old embarks upon a life of crime and costs society 2.7 million to 4.8 million dollars to navigate through the criminal justice system rather than matriculating, obtaining a college degree, and contributing as a tax paying citizen (Cohen & Piquero, 2007).

Research Questions

The following two key research questions will guide the study:

- 1. What are the elements of quality out-of-school time programs for school age youth as perceived by the participants?
- 2. What is the impact of quality as perceived by administrators, teachers, and students in out-of-school time programs?

Research design. The purpose of this research design is to provide a framework for this case study based on the premise that quality in out-of-school time programs matter. This study is designed around the premise that OST programs built around key elements of quality reinforce outcomes of student achievement and personal enrichment. A qualitative approach provides the best pathway for this researcher to explore the effects of quality keys in OST programs as related through the experience of all participants: program administrators, faculty (site coordinator, teachers, paraprofessionals, and peer tutors), students, parents and community partners. Descriptive research designed around a case study "focuses on understanding the dynamics present within a single setting" (Eisenhardt, 1989, p. 534).

Research setting. The following is a description of a middle school out-of-school time program in Southeast Arkansas. This OST program strives to assist poor and minority students

with academic challenges. The program begins at 3:00 p.m. as 80 – 130 students start the program with a nutritional snack and then turn to an emphasis on academics. Certified teachers, classroom aides, and peer tutors work with students to complete homework assignments and receive one-on-one help to increase proficiency in literacy and math. The program ends at 6:00 p.m. and includes time for student enrichment and recreational activities. Students choose enrichment and recreation activities among offerings such as science lab, arts and crafts, music classes (guitar, piano), physical education, drug prevention programs emphasizing alcohol and tobacco abstinence, archery club, book club, nutrition and cooking class, character education, service learning projects, technology camps, foreign language class, and virtual field trips.

Middle school teachers reported students participating in the after-school program completed homework, improved their grades, and experienced fewer disciplinary problems in the classroom. This school has experienced improvement in attendance rates and standardized test scores. In the 2008-09 school year, African-American students scored sixty-seven percent (67%) in math and sixty-four percent (64%) in literacy compared to overall student scores on the benchmark exam of eighty-six percent (86%) in math and seventy-nine percent (79%) in literacy. African-American students and students from poverty stricken families are still behind other students but show marked improvement on standardized test scores (Beebe, 2008).

Theoretical Framework

Maslow (1954) constructed a theory derived primarily from his clinical experience as a psychologist, and secondarily as an effort to synergize the knowledge of functionalists, James and Dewey, the holistic psychology of Wertheimer, Goldstein, and Gestalt, and the dynamic work of Freud, Fromm, Horney, Reich, Jung, and Adler. Maslow's theory is often over-

simplified as a step-system of human gratification of needs; however, the reality of his theory represents a complex synergy of modern era knowledge that endures into the post-modern age and beyond. The theorist concluded that reaching the highest level of human motivation (self-actualization) is only possible when other basic human needs are met. Maslow explained this actualization process as reaching ones full potential, that is, "to become everything one is capable of becoming" (p. 22). Self-actualization will vary from person to person depending on individual desires; however, there is one thing common to all cases, other basic needs: "physiological, safety, love, and esteem needs" have been already met (p. 22). The model (Figure 1) is linked to Maslow's hierarchy and suggests key elements to build a quality framework that must be present in out-of-school time programs.



Figure 1. Quality elements for out-of-school time programs

Limitation of the Study

The limitation of the case study included the site selection of one 21st Century Community Learning Centers middle school located in Southeast Arkansas rather than multiple sites with similar demographics. The selection process included a middle school funded by the 21st Century Community Learning Centers authorization and identified by the Arkansas Department of Education as under "School Improvement." As a result of the requirements of programs funded by 21st Century Community Learning Centers, the target population included 50% or more economically disadvantaged students with three or more risk factors. The risk factors included single-parent families, limited English proficiency, and academic failure. Students selected for the out-of-school time program were identified based on test scores, academic performance and the need for enrichment activities beyond the regular school day. Because of the risk factors, students participating in the out-of-school time program were representative of certain subgroups. Restrictions to the OST program due to location, program schedule and district policies also limited the study.

Definition of Terms

The following definitions are listed and summarized below to provide an understanding of the terminology used in the study (Arkansas Department of Education, 2010).

21st Century Community Learning Centers (21st CCLC): a community learning center located within a school district, nonprofit or faith-based organization that offers academic, artistic, and cultural enrichment opportunities to school age students and their families during non-school hours. The centers are administered through state agencies under the reauthorization of No Child Left Behind Act of 2001 (NCLB, 2002).

Academic Achievement: refers to the success of students in learning and mastering the school subjects that they study as measured by tests of knowledge and skills.

Adequate Yearly Progress: a measurement used by the U.S. Department of Education that is included in the No Child Left Behind Act of 2001(NCLB, 2002) to determine how school districts are performing academically as shown by the results on standardized tests in mathematics and English Language Arts. Test results are determined for whole groups as well as for subpopulations of students.

After-School Programs: activities following the official end of the school day, typically sponsored by the school, district, or community organization. Interchangeable terms with after-school include, school age care, out-of-school time and expanded learning opportunities.

Alert Status: schools are designated this status by the Arkansas Department of Education the first year the school fails to meet adequate yearly progress.

At-risk Students: students identified and are in danger of failing in school and becoming academically disadvantaged in comparison to peers.

Latchkey Children: children that are left in homes without adult supervision during non-school hours.

No Child Left Behind Act of 2001: No Child Left Behind (*formerly called the Elementary and Secondary Education Act*) is a set of laws that supports standards based education and calls for disaggregation of student-performance data by student subgroups.

Out-of-school time (OST): refers to both traditional after-school programs operating in the afternoon hours as well more comprehensive programs that may include weekends, summers and

holidays. Programs may be offered by schools, community and faith based organizations, libraries, museums, municipalities, and volunteer groups.

Quality in out-of-school time programs: refers to the key elements that constitute program quality and promote positive youth development outcomes. Key elements include a safe and appropriate environment, training, evaluation, youth development, parental involvement, attendance and participation.

School Improvement: school designation due to the failure of the school to make adequate yearly progress for two or more consecutive years.

Conclusion

Chapter one provides the introduction, background, statement of the problem, purpose and significance of the study. The research questions to be answered, theoretical framework, limitation of the study, definitions of terms, and conclusion are also defined in Chapter one. Chapter two presents a review of related literature which includes the background, methods, search strategy, historical perspective, and impact of OST programs. The literature review includes the theoretical framework, significance, summary and conclusion. Chapter three includes an overview of the methodology, research questions, researcher's role, data management and summary. Chapter four presents the data analysis, introduction, description of participants, data management, research questions, elements of quality, description and distribution of themes and categories and summary of data. Chapter five includes the conclusion and recommendations, introduction, summary, research questions, interpretation of data, program and field recommendations. Chapter five concludes with the recommendations for future research and the conclusion.

Chapter Two

Review of the Literature

Changes in American society since the industrial revolution have presented the need for quality out-of-school time care for children and youth. The changing family structure caused by both parents working outside the home, the advent of the single-parent household, the necessity of federally mandated standardized testing for student achievement, criminal activity and the expanding population of children, has contributed to the problem of how to protect children and enrich their lives during out-of-school time hours. Equally, Fashola's (2002) study strongly suggests the interest in out-of-school time (OST) programming encompasses societal issues such as welfare reform, changes in family structure, increased juvenile crime, and the overall poor academic performance of students. According to Miller (2003), the purpose of school or community based OST programs is to help working parents balance family responsibilities and promote "social, emotional, creative and physical development" of children in various activities (p. 25).

Therefore, the purpose of this study is to describe key elements of quality in out-ofschool time programs for school age youth as perceived by the participants. The study will identify standards for high quality out-of-school time programs linked to student achievement and positive outcomes for participants. Thus, the following two research questions will guide the study:

(1) What are the elements of quality out-of-school time programs for school age youth as perceived by the participants? (2) What is the impact of quality out-of-school time programs as perceived by administrators, teachers, and students?

The aim of the literature review is to provide educators, policymakers and researchers with a review of research related to quality standards and youth development that is evaluative, comprehensive and current. The review of literature is categorized into nine major components: (1) background (2) methods (3) search strategy (4) historical perspective (5) impact of OST programs (6) theoretical framework (7) significance (8) summary, and (9) conclusion.

The review includes a discussion of youth development, quality standards and existing research on the relationship between out-of-school time programs and student achievement. While the review was narrowed to focus on out-of-school time programs that include measures of academic improvements, other benefits were addressed and included in the review. These benefits are related to the safety, social and developmental needs of youth.

Background

Improving student achievement and promoting the growth and development of youth are two of the major goals of out-of-school time programs. However, the growing population in the United States expands the problem facing parents, educators, and policymakers of how to manage children's out-of-school time.

According to the U.S. Census (2009) more than 58 million (58,528,070) students were enrolled in school including nursery school, preschool, kindergarten, elementary and secondary school programs in the United States in 2008. The 2008 statistics depict an increase of more than a half million students (653,263) in student enrollment over 2002 which was over 57 million (57,874,807). The dramatic increase of children in the U.S. population is illustrated by looking at population increases of school age children from the year 2000 to projected populations of children by the year 2050. In 2000, children ages 6-11 total 25 million and children ages 12-17 total 24.2 million. By 2050, children ages 6-11 expand to 34 million and children ages 12-17 grow to 33.8 million, which projects a marked increase of school age children. By 2050, there will be an increase of children ages 6-11 of 9 million and an increase of children ages 12-17 of 9.6 million.

Out-of-school time programs benefit children and youth by increasing safety during outof-school time hours, reducing risk-taking behavior, and improving learning. There is an assumption during out-of-school time hours, juvenile crime peaks and out-of-school time programs provide a safe, structured, supervised, and academically engaging place for students and thereby reduce the temptation to engage in risk-taking behaviors related to the use of drugs, alcohol and tobacco. Keeping children safe and active in an out-of-school time program prevents crime, juvenile delinquency, and violent victimization. The positive impact of out-ofschool time programs continues to be validated by research and the findings supported by the Arkansas Out of School Network (2010) report entitled, *Afterschool is Key* and Arkansas Advocates for Children and Families.

According to a study prepared for The After-School Corporation (TASC) and the Southwest Educational Development Laboratory (SEDL) by Policy Studies Associates (2005), quality out-of-school time programs make a real difference not only in academic achievement for participants, but also provide a support system and offer a variety of youth oriented activities. As a result, a review of literature is conducted in order to gain an understanding of out-of-school time in correlation with the quality standards. The purpose is to review empirical studies related to out-of-school time programs which include an academic measure of performance and reflect the needs and characteristics of the students and community. The goal of the case study is to describe the key elements of out-of-school time programs for school age youth. The second goal is to determine the impact of the quality standards in out-of-school time programs as perceived by the administrators, teachers, and students. The participants for the study are middle school students attending a 21st Century Community Learning Center after-school program in Southeast Arkansas. Questions include the following: what programs and services are needed to address the needs of low performing students? What is the relationship between 21st Century Community Learning Centers after-school programs and the regular school day? What are the characteristics that define quality in out-of-school time programs?

As a final point, according to Miller (2003) out-of-school time programs must identify and address the unique needs of the students and parents within the community through a variety of activities and opportunities. Out-of-school time programs play a vital role in the academic achievement, growth and development of youth in school districts. However, the structure, personnel, programs and funding are key elements that must be addressed in order to provide high quality out-of-school time programs. The focus is centered on youth development and student achievement; however, the programs must be more than an extension of the school day.

Methods

The studies included in this review of literature were based on four general criteria: (a) relevance; (b) quality; (c) empirical; and (d) scholarly nature. To evaluate the relevance of a study, the researcher determined if it was applicable to the research questions and addressed quality in relationship to out-of-school time programs. Due to the historical significance of out-

of-school time programs in relationship to the safety and welfare of school age children, older studies are included in the research and correlated with current research to reflect the changes in family composition and economic issues affecting children and families.

The criterion for the selection of qualitative and quantitative studies was determined by relevance, findings, sample size, validity, and rigor. Publications including books, book chapters, monographs and journals that failed to provide evidence to support the conclusion were restricted. In addition, the review included professional organizations and legislation relevant to the study.

Search Strategy

The search process utilized the electronic database that included ERIC, Google Scholar, ProQuest, Education Abstracts, Policy Briefs, educational journals, dissertations and books. The 21st Century Community Learning Centers national and state offices provided relevant and pertinent information regarding efficacy of out-of-school time programs. Due to the vast number of studies and articles, the criteria for selection and the two key research questions were the determining factors for the inclusion or exclusion of studies. The timeframe includes a brief history, from 1928 to the 2011 legislation.

Historical Perspective

After-school programs were started in the nineteenth century by community and faithbased organizations to care for children during out-of-school time. Halpern (2002) reports findings cited by Brenzel, et al. (1985) regarding the time education became compulsory for children in 1928 school enrollment increased by 80 percent as the labor force for children declined. The need for childcare was eminent by 1940 due to the growth of employment opportunities for females which contributed to the demographic shift in the United States (Halpern, 2000). Demographic shifts contributing to the need for out-of-school time programs include the baby boom period, growth of single parent homes, and a lack of extended family members to care for children during out-of-school time. As a result, the government increased its support for after-school programs. Increasing societal concerns are creating a growth in the after-school movement related to the number of working households, extending learning opportunities for low-performing students and the increase in juvenile crime during non-school hours (Kugler, 2001).

According to Halpern (2002), the role and importance of out-of-school time programs was defined by historical events dating back to the nineteenth century. The historical events included (a) Defining the purpose and role of OST programs; (b) The Great Depression and World War II; (c) Elementary and Secondary Act (ESEA) of 1965; (d) Reauthorization of ESEA in 1966; and (e) No Child Left Behind Act of 2001 (NCLB, 2002). The timeline following these historical events impacted the economy, families, and the number of women entering the workforce.

Richard Riley's speech to grantees at the 21st Century Community Learning Centers Summer Institute 2000 further defined the purpose and role of out-of-school time programs:

Each weekday afternoon in America, the ringing of the school bell signals not just the end of the school day, but the beginning of a time when at least 8 million of our children are left alone and unsupervised. For working parents, ensuring appropriate supervision of their children during the afternoon can be an extremely difficult challenge. As a result, so called "latch-key" youngsters can be found in our urban, suburban and rural communities where working parents, for a variety of reasons, are unable to arrange or afford a better alternative. Instead of being a time for growth and opportunity for these children, the hours immediately following the school day are their most dangerous, for these are the hours when children are most likely to commit or be the victim of crime. For many others, the afternoon hours are simply a period of idle and wasted time, when opportunities to be monitored and academically challenged are squandered. (2000, p. 1)

In conclusion, Riley's speech addressed the need for out-school time programs and supported the benefits of these programs. By addressing the need to keep children safe, support working parents and provide opportunities for children to explore and learn in a quality setting, out-of-school time programs provide a safety net, eliminate unproductive time and support the growth and development of youth.

No child left behind act of 2001. The No Child Left Behind Act of 2001 (NCLB, 2002) requires high-poverty schools to show progress in standardized test scores in English and math, but disregards many aspects of student's experiences and goals of public school (Ascher, 2006). The NCLB Act requires students to "reach or exceed each state's proficient level of performance in reading and mathematics by the end of the 2013-2014 school year." States must report disaggregated scores for "economically disadvantaged students, students with disabilities, students from major racial and ethnic groups, and students with limited English proficiency" to mark progress of student subgroups (p. 3). Educators create supplemental programs to address the needs of students who are performing below basic on standardized tests including out-of-school time programs (Miller, Snow, & Lauer, 2004).

Out-of-school time programs. In 1998, the largest federally funded program under the No Child Left Behind Act of 2001 (NCLB, 2002) the 21st Century Community Learning Centers authorized under Title X, Part 1, of the Elementary and Secondary Education Act, was introduced redefining the role of out-of-school time for school age children. Out-of-school time programs have taken on a new role in public education to bring opportunities for academic remediation and enrichment activities to students in need of support and mentoring Miller et al. (2004). One of the intended results of out-of-school time (OST) programs is for students to become proficient on standardized tests.

The U.S. Department of Education classifies the 21st CCLC as a key component of the No Child Left Behind Act of 2001 (NCLB, 2002). The 21st CCLC Program directive is to "establish or expand community learning centers that operate during out-of-school hours (before school, after school, or during holidays or summer) or full-day Prekindergarten programs" for four-year-olds. 21st Century Community Learning Centers provide opportunities for "academic, artistic, and cultural enrichment." The program statute stipulates in section 4201 (1) (B) that a community-learning center "assists students in meeting state and local academic achievement standards in core academic subjects, such as reading and mathematics through academic enrichment opportunities along with other activities designed to complement the students' regular academic program." Families of students targeted for the program must be offered "literacy and related educational services" (United States Department of Education, EC 4201, 2002b, p. 2).

OST program evaluations. In 1999, the U.S. Department of Education contracted with Mathematics Policy Research, Inc., and Decision Information Resources, Inc. to complete an evaluation of 21st Century Community Learning Centers program. When schools stay open late: The national evaluation of the 21st Century Community Learning Centers program is considered to be the largest and most rigorous examination of school based out-of-time programs (U.S. Department of Education, 2003b). Student outcome data were collected in 21st CCLC elementary and middle school programs. The data were categorized and linked to student and school outcomes. The five categories were: (1) after-school supervision, (2) location and activities, (3) academic performance and achievement, (4) behavior, and (5) personal development and safety. Findings of the first year report were released in 2003 and indicated that programs failed to provide academic improvement in math and literacy. The program evaluation was also unable to determine whether participants felt safe in the program environment. No significant changes were noted in the areas of behavior, interpersonal skills, parental involvement, self care and supervision. Year two findings reported data that were consistent with year one. Elementary students test scores and grades failed to show measurable improvement. The report indicated middle school students' grades were higher in social studies; however, lack of improvement was noted in English, math, and science. The report further indicated elementary students reported positive feelings in the area of safety, unlike middle school students reporting negative feelings in program safety. Minimal impact was noted on parental involvement for elementary students as well as middle school students. The annual evaluations are continuous for 21st Century Community Learning Center programs.

Huang, Gribbons, Kim, Lee, and Baker (2000) completed a longitudinal study on LA's Best after School Enrichment Program for K-5 students. The purpose of the study was to analyze the program results of the students enrolled in the program and to compare the results with non-participants. Background information obtained on both groups included ethnicity, gender, disability, economic status, test scores and the number of years students were enrolled in the program.

The findings of the LA's Best After-School Enrichment Program were positive based on student achievement and student performance. However, the findings indicated there were changes in the standardized test used by the district. The report indicated the school district changed from administering the Comprehensive Test of Basic Skills (CTBS) to the Stanford-9. Later, the district changed to the SAT-9, Form S and then to SAT-9, Form T. The findings of the LA's Best study indicated participants felt safe in the after-school program and parents were positive regarding safety. Participants in LA's Best were more engaged in school as reflected by improved attitudes and positive relationships with adults and other students. Parents also expressed high expectations for their children. The children exhibited a personal interest in their future as demonstrated by answers on a questionnaire. The academic performance of the participants improved by a letter grade in the core subject areas. This was correlated to the number of years the students were enrolled in the program. The report concluded by discussing the economic factors that continue to affect the living conditions of students with an increasing number who are living in poverty and have a need for the program. According to Huang et al. (2000), "The rationale for LA's BEST and its programs are even more important and necessary today than they were twelve years ago, when LA's BEST was created" (p. 21).
Kane's (2004) study, in a working paper supported by the William T. Grant Foundation endorses the argument after-school programs are not only being defined as a new institution but are also part of the "national policy debate" (p. 1). Kane's report summarized the results of the four studies that include the 21st Century Community Learning Centers (2lst CCLC). The commonality among the four studies included centers that operated on a voluntary attendance basis typically averaging four or five days per week. Second, information was limited with only one study addressing the out-of-school time arrangements for participants in comparison to nonparticipants. Third and most significant of the results of the evaluations of the studies questioned the long term academic performance and impact of the students that remain in the program. Fourth, the findings indicate out-of-school time programs increase parental involvement in school, student engagement, and promote higher interest in the completion of homework, which was consistent in the majority of the program evaluations. The role of out-of-school time program and the expectations according to Halpern (2002) is often to function in a family and school support role by addressing academic and social needs that impact program goals and the identity of the programs.

Impact of OST Programs

Fashola's (2002) work on the impact of OST programs on at-risk youth in educational settings began in the 1990's. Fashola acknowledged the "problem of the after-school movement is the breath of its potential outcomes" (p. ix). Designing quality OST programs may require limiting program focus to address more select outcomes for the children and youth it serves. Funding sources help establish outcomes by defining evaluation measures such as standardized test scores in math and literacy. At-risk students are usually identified by performing below

basic in these areas. However, due to funding guidelines these students are mixed with proficient students in OST programs with less focus on academic achievement program outcomes.

Fashola (2002) evaluated thirty-four out-of-school time programs meeting four categories of selection criteria. First, the out-of-school time programs address a specific academic component, specific curriculum area, tutorial program with a focus on reading improvement and may be identified as a community-based program located within a school. The second category recognized that professional development and staff training helps ensure program success. Programs should rely only on qualified or certified staff to provide academic instruction. The third category recognized the importance of using pre-post data and school attendance as a measure of program success. The fourth category involved identifying barriers to student participation regardless of program location. For any program to be successful, the researcher concluded the categories identified must be in place to track the progress of the program and to successfully sustain the program.

Vandell, Riesner, and Pierce, (2007) conducted a two-year longitudinal study of the effects of high quality after-school programs tracking 3,000 low-income and ethnically diverse elementary and middle school students across eight states in urban and rural settings. This study determined regular participation in quality programs corresponds to increases in standardized test scores, improvement in work habits, and decreases in negative behaviors of disadvantaged children and youth. Vandell et al. (2007) constructed a theoretical model demonstrating the affect of students' personal and family background and developmental level prior to enrollment in an afterschool program. Program inputs such as correct dosage of program elements are

combined with experience in promising programs, sports activities, academic and enrichment activities, adult supervision at home, and supervised activities. Program outputs produce intermediate and long term outcomes including: (1) improved interpersonal behavior and social skills, (2) improved grades and work habits, (3) improved test scores, and (4) reduction in risky behaviors and disciplinary sanctions.

Vandell et al. (2007) reported the promising programs studied had strong community, school, and neighborhood partnerships in place in the communities they served. High quality programs were assessed using a rating system to document evidence of supportive interpersonal relationships between staff and students, evidence of academic support and enrichment activities such as recreation, opportunity to explore the arts, and other enrichment activities. The correct mix of enrichment activities were noted to build interpersonal relationships between students and to keep them actively engaged in activities. Promising programs offered age-appropriate opportunities for learning, tutoring, and games to enhance reading and math skills, recreational activities, community service, art and other enriching experiences. These programs provided staff training, maintained low student-to-staff ratios, and provided a strong partnership with schools and parents.

The findings acknowledged benefits of regular engagement in high-quality out-of-school time programs, community activities, and adult supervision at home for economically disadvantaged students. Risk factors for student participating in unstructured extra-curricular activities combined with minimal supervision at home during out-of-school time hours were identified. The risk factors included lack of adult supervision, boredom, engagement in risk taking behaviors which may lead to criminal activities.

Further research by Bodilly and Beckett (2005) identified three categories of literature pertaining to common out-of-school time program elements implemented to produce positive program outcomes: (1) school-age care; (2) youth-development; and (3) education literature. All literature types acknowledge that physically and psychologically safe and well organized environments are an important part of quality in out-of-school time programs. Youth-development and education literatures acknowledge high expectations in the areas of conduct, learning, and achievement as important to the quality out-of-school time program outcomes. School-age care and youth-development literatures acknowledge the need to offer age-appropriate and challenging opportunities for participants to learn a new skill and the importance of implementing sustainable parental, community, and volunteer partnerships to support youth in quality OST programs: (1) limitations on program and classroom size, and (2) teacher and staff training and the importance of clear program objectives with frequent assessments of whether the program is meeting objectives.

Findings from the study by Bodilly and Beckett (2005) on out-of-school time evaluations focus primarily on academic achievement rather than child-care arrangements. Impacts of OST programs focused on four specific areas: (1) health & safety, (2) attainment measured by grades, (3) social and health behaviors, and (4) social interactions. Bodilly and Beckett (2005) noted the evaluation was program specific and failed to account for the impact of participant involvement and overall participant engagement. The four impacts of out-of-school time programs identified by the researchers were recognized as elements of quality but failed to address participant involvement involvement in relationship to these particular elements of quality.

Similarly, Baker, Speilberger, Lockaby and Guterman (2010) report practitioners and policymakers have recognized three program elements that have the most effects: (1) programs addressing multiple developmental domains; (2) high quality programs, and (3) professional staff successful in sustained engagement of children. Baker et al. (2010) determined the challenges of improving quality revolves around unstable funding and staffing conditions, and the ability to establish realistic program objectives and quality standards for a diverse field of after-school program providers. The field has recently embarked on a mission to create and implement a system of quality standards, supports, and resources for out-of-school time programs that will be effective regardless of program type or location.

Implementing high quality, according to Baker et al. (2010) is more difficult when key program elements supporting quality outcomes are not sustainable such as: (1) inadequate funding; (2) unqualified staff and high staff turnover; and (3) inadequate space. Start-up organizations with low levels of quality have difficulty making quality improvements because there is a perception that program operation is overwhelming and attempts at improvements would be unsustainable.

Collectively, studies show that participation in high quality programs produce positive outcomes. There is no stream-lined formula for success to enhance program quality; however there are common characteristics including: (1) maintaining trained, caring staff; (2) programming that is culturally and developmentally appropriate that addresses youth's interests and needs; (3) facilities that are designed around safety and accessibility with adequate equipment and materials; (4) programs that have attended to health and nutrition needs; (5) programs with strong management and administration with effective community and school

partnerships, and family engagement; and (6) programs with ongoing evaluation, goal setting, and improvement (Children Now, 2001; Eccles & Appleton, 2002; National School-Age Care Alliance, 1998; Peter, 2002).

Theoretical Framework

Out-of-school time programs play a vital role in the growth and development of children. The growth and youth development factors are correlated to Maslow's (1943) Hierarchy of Needs in relationship to the developmental needs of youth in out-of-school time programs. Maslow's Hierarchy of Needs developed by U.S. Psychologist Abraham Maslow in the latter half of the 20th century describes human behavior in relation to the basic and higher needs people experience in their lives. Similarly, the core commitments of the Arkansas Standards for Quality After-school Programs addressed the critical areas necessary to meet the developmental needs of children and youth for the first two decades of their lives by taking a positive approach to ensure that all children have access to high-quality development opportunities during out-ofschool time (Arkansas Out-of-School-Network, 2009).

Maslow's (1943) Hierarchy of Needs defines the greatest and lowest needs at the bottom of a pyramid and the highest needs at the top in a given order. The five needs identified by Maslow are physiological, safety, love/belonging, self-esteem and self-actualization. The physiological needs that are most familiar represent the basic needs for survival that include food, clothing, shelter and sexual activity. According to Maslow, these needs must be met in order for the body to function. Therefore, out-of-school time programs address a critical need by providing programs that promote healthy lifestyles and nutrition as a part of the activity schedule. The safety needs are the second layer of human needs identified by Maslow upon satisfying the physical needs. The safety needs are individualized in relationship to feeling safe and secure within one's environment. Order, consistency, health and well-being are linked to the safety and security needs. The Arkansas Out of School Network (2009) quality standards identified a safe environment with stability and adult supervision as one of the priority needs for youth in a quality program. The youth needs list also includes food, shelter, healthy choices, and health prevention. In contrast, the number one element listed under the quality standards is safe and appropriate program environments and facilities to ensure the safety and welfare of children. The third human need identified by Maslow is the social need which includes friendship, intimacy and a support system. The need to belong and to be accepted or to be a part of a group/team is often fulfilled by the out-of-school time program. The researcher further discussed a lower and a higher version of esteem needs related to the need for recognition in the lower tier and self-respect in the higher tier which is the fourth layer of the hierarchy.

Improving academic attainment is paramount to success throughout the life course of an individual and becomes the highest goal of quality out-of-school time programs. Instilling value of an education and the benefits of postsecondary education changes the outcome of student lives. The potential life impact for a student realizing the goal of academic attainment can be compared to the life impact of an individual reaching their full and true potential in Maslow's (1943, 1954) hierarchy of human motivation. The assumption is that educational attainment allows individuals to move up in the hierarchy of society, it is the way out of a life based on struggling to achieve the lower needs of basic physiological and safety needs.

Descriptive elements of program quality. An out-of-school time program measures its success through monitoring and evaluating improvements in academic achievement of participants. Successful programs become sustainable by developing key components of effective leadership, continuous program evaluation, multiple funding sources, and vested student, parent, school, and community stakeholders. Once a program is sustainable, other support structures can be put in place that will produce positive impacts on children and youth through encouraging regular program participation.

Subsequently, Miller (2003) adapts the theory of change to depict effective OST program features and program outputs. External environmental contexts of family, school and community together with internal contextual factors of race and ethnicity, temperament, and personality of students are inputted into program features including: (a) partnerships combining supportive efforts of family, school, and community; (b) appropriate structure and provision for physical and psychological safety needs; (c) opportunities to build and support interpersonal relationships; (d) opportunities to attain sense of belonging, efficacy, and mattering; (e) opportunities to attain new skill sets. The efficacy of program features directly affects students and provides direct program results such as: (1) students benefiting from increased involvement with family; (2) students benefiting from caring adult and mentor relationships; (3) students benefiting from new sense of belonging to a positive peer group; (4) students benefiting from a chance to practice skills and accrue new knowledge; and (6) students gaining self-awareness of academic competence.

Direct program results produce positive program outputs such as increased school engagement and therefore, increased school achievement. Increased school engagement includes

30

the following factors: (a) students are more motivated to do well in school; (b) students have higher attendance; (c) students build better work habits and are more persistent; (d) students behave better in the classroom; and (e) students increase cognitive skills. Increased school engagement sets the stage for increased school achievement in three areas, including: (1) higher test scores and grades; (2) lower incidence of students repeating a grade; and (3) higher graduation rates.

Miller (2003) states program evaluation based on the theory of change form a basis to examine the links between how the program works and how students are affected, but, outcomes cannot be attributed to program participation scientifically. However, on-going studies are examining the potential of theory-based evaluations because they require fewer resources. According to Vandell et al. (2007), theory based evaluation identifies the strengths and weaknesses of a program model and target the focus of the evaluator on the problem areas. The evaluation also looks at the results or expected outcomes based on program participation which may be defined by the funding source or a school district.

States' quality standards. According to the U.S. Department of Health and Human Services Administration for Children and Families (2010) there are twenty-nine states that have established or in the process of establishing school-age care program and practitioner quality standards that rise above basic licensing regulations. Seventeen states have developed schoolage quality standards for programs addressing curricula, program administration, outreach to parents, and learning environment. Eighteen states have practitioner standards; fifteen states have core competency standards; thirteen states have credentialing standards for school age children. Table 1 list the states with school-age care program and practitioner quality standards. .

Table 1

| School-Age Program Standards | Core Competency Standards | School-Age Credential | | | |
|------------------------------|---------------------------|---------------------------|--|--|--|
| California | California | Colorado | | | |
| Connecticut | Delaware | Delaware | | | |
| Delaware | Georgia | Indiana | | | |
| Florida | Indiana | Kentucky Massachusetts | | | |
| Indiana | Kansas | Michigan | | | |
| Iowa | Massachusetts | Missouri | | | |
| Kansas | Missouri | New Hampshire | | | |
| Maryland | Nebraska | New York | | | |
| Minnesota | New Hampshire | Ohio | | | |
| Missouri | Ohio | Pennsylvania | | | |
| New York | Oklahoma | South Dakota | | | |
| North Carolina | Oregon | Wisconsin | | | |
| Ohio | Rhode Island | | | | |
| Oregon | Vermont | | | | |
| Rhode Island | Washington | | | | |
| West Virginia | | | | | |
| Wisconsin | | | | | |

Quality Standards for School-Age Care by States

Existing national and local OST standards raise the following core issues: (1) program management and administration including program organization, policies, planning, fundraising, fiscal management, and supervision; (2) program activity including flexible scheduling to address needs of children and youth that offers security, independence, variety, and stimulation; interpersonal relationships that recognize the nature of interactions among youth, families, staff, other stakeholders, and staff to child ratios; and (3) characteristics of safe indoor and outdoor learning environments with equipment and materials that will engage children and youth; program highlights that address health and nutrition needs of children and youth.

Other quality standards frequently appear in the literature but were not consistent across programs including: (1) youth engagement and leadership; (2) family participation and

engagement; (3) program evaluation criteria; and (4) child and youth development activities (Children Now, 2001; Tolman et al., 2002). Youth engagement and leadership are often neglected in the program development process but will ensure positive program participation and outcome by children and youth and should be implemented into quality standards (Eccles & Appleton, 2002).

Quality standards establish program consistency; enhance program quality, and increase program recognition and potential funding. Harvard Business School assisted the organizations, Rhode Island Kids Count and Community Matters, in reviewing best practices and impacts of quality indicators for OST programs. National, urban, and local OST programs were considered in the findings. Developing OST standards provides a framework for systems of program improvement strategies. Quality standards that include both program standards and practitioner competencies make providers, families, funders, and other stakeholders aware of which strategies promote positive youth outcomes (Community Matters & Breslin, 2003).

Significance

The over-arching purpose of an out-of-school time program is to reduce the barriers to student achievement by providing additional instruction time for struggling students, tutoring, and help with homework in a safe, supervised environment geared for learning enhancement. Barriers of student achievement include generational poverty, low parental and community educational attainment, and lack of employment opportunities.

The 21st Century Learning Centers program has become the largest federal funding source for out-of-school time programs in the United States. This program was a major initiative of the Clinton-Gore Administration to keep children safe and provide meaningful enrichment during out-of-school time. Former Vice-President Al Gore recognized children were most vulnerable during the time school lets out until parents return home from work because children and youth are most likely to engage in at-risk behavior while unsupervised and may be in unsafe environments.

Out-of-school time programs are important as reflected by the 2008 polling conducted by the Lake Research Partners on out-of-school time programs where nearly nine in ten voters (89%) noted the danger young people face, which further highlighted the significance of after-school programs. The consensus among voters was that out-of-school time programs play an important role in keeping youth in school; provide a safe place and opportunities to learn during the peak hours from 3-6 p.m.

The America After 3 PM (2009) study sponsored by the JC Penney After-school Fund provided current information regarding how America's children spend their afternoon. The national findings of the study reported "children who are unsupervised after-school are not only in danger of becoming victims of crimes or accidents; they are also at risk in other less dramatic but equally troubling ways" (Afterschool Alliance, 2004, p. 1).

In the Arkansas After 3PM (2009) study of 505 households surveyed for the study, 26% (125,025) of Arkansas' K-12 children were unsupervised after-school. The study further report 44% (187,722) of Arkansas parents cited the lack of availability of out-of-school time programs. The children would participate if programs were available regardless of their current care arrangement (Afterschool Alliance, 2004).

Baker et al. (2010) reports the after-school field has grown considerably over the past two decades and has acknowledged the existence of achievement gaps between low-income and more

economically advantaged students and has sustained the belief that after-school programming can help reduce these gaps. The achievement gap between white and minority students continues to rank as a high priority in school districts and a major public concern.

Barth and Nitta (2008) suggest providing more instructional time for low-performing students is one of the factors in closing the achievement gap by supplementing the regular school day. The needs and benefits of after-school programs are higher for lower-income students than middle-income children (Miller, 2003).

Summary of Review of Literature

The review of literature provides a synopsis and program evaluation of historical and current research on quality standards and out-of-school time programs for school age youth. Some of the findings were consistent with the researcher's as outlined:

Fashola (2002) concluded as out-of-school time programs work toward their goals, it is critical for programs to track their progress and report results. Program evaluation significantly influences policy, administration, education, and research. Fashola (2002) noted the need for evaluation of these and other current OST programs to produce effective and replicable programs for increasing student achievement and positive outcomes. Clearly more work on the quality front in OST programming is necessary to replicate programs that show evaluative results.

Baker et al. (2010) reported program stakeholders perceived the concept of quality differently because the meaning and importance of quality vary across programs with differing interests and objectives. Thus, the concept of quality is not clearly understood and there is no understanding of how to achieve quality, the benefits quality brings to organizations that attain it, or the necessary program inputs to achieve quality such as funding support and other resources. Quality standards become a tool to establish common professional terminology used by practitioners in child care and developmental programs for school age children, families, schools, funders, policy makers, and other stakeholders and useful tools for goal setting, program evaluation, and improvement. Evaluating program effectiveness leads to sustainable community partnerships, funding, staff, public recognition, and program participation. Quality standards hold programs accountable for outcomes and shortfalls and provide potential funders a basis for investment in proven practices. Standards are foundational to sustain staff and development efforts by linking quality keys to training and technical assistance. The use of recognized quality standards provide a mechanism to coordinate conflicting methods of assessment used by various funders and policy makers. Quality standards earmark the cost of quality in programming and can guide how program funds are distributed (Hall, 2002; Tolman et al., 2002).

Bodilly and Beckett (2005) call for "more rigorous assessment" to identify OST program components that provide the most benefit to participants, but caution against isolating program features for evaluation because of cost and complexity (p. 74). The literature review led researchers to determine that quality may be engineered into OST programs by designing around program features recognized as elements of quality. Programs demonstrating effectiveness can be closely evaluated to determine how program components relate to overall program quality. A cost analysis of effective programs can identify the right mix of program components that produce desired outcomes.

Further, in reviewing the Bodilly and Beckett (2005) study, it became apparent to this researcher that the literature lacks focus on participant's voice and choice in quality OST programs and lack of measurement or reporting regarding participant input into quality

programs. A measure of quality in OST programs may be linked to the participant rather than program features.

Also, Little, Wimer, and Weiss (2008) Issues and Opportunities in Out-of-School Time Evaluation briefs summary of ten years of research on after-school programs concluded the increasing growth in funding and participation in after-school programs provoked further discussion related to benefits, outcomes, and accountability. However, the common theme in a series of studies under the Harvard Family Research Project affirms the need for and impact of quality standards in after-school programs regardless of the location and type of program which is supported by Governor Mike Beebe (2008) and the Arkansas Out-of-School Network (2010).

The final report of Beebe's (2008) Task Force for after-school and summer programs defined out-of-school time programs based on a framework of quality standards and best practices that include a comprehensive system approach and accountability to ensure that out-of-school time programs are addressing the needs of participants. The framework for quality standards and the key elements included in the report identified program expansion, evaluation, and training for out-of-school time staff. The expansion process included the utilization of existing agencies by expanding their capacity to promote quality standards and administer programs. The final step required the use of research-based practices to help determine the number of hours and days per week for students to produce positive outcomes. Identifying elements of quality in OST programs is difficult because of the differences in organizations and their program objectives, goals, and desired program outcomes, their populations served, their partners and other stakeholders (Community Matters & Breslin, 2003).

Conclusion

The review of literature did not identify a streamlined formula for creating quality out-ofschool time programs or a theoretic model that addresses quality standards for different program types cross various geographic locations with the common goal of increasing academic achievement and other positive enrichment and developmental goals for students. The differences of program types and stakeholders make it difficult to align quality elements in outof-school time programs. Elements of quality were identified in the literature and many successful out-of-school time programs utilize many of these elements. Some programs have implemented all the elements while others overlook key elements that would make their program more sustainable. The overall goal of out-of-school time programs is to create student success whether through academic achievement, personal development, or enrichment activities.

Additional studies also addressed the social barriers and unmet needs defined by Maslow's Hierarchy of Needs. The gaps acknowledged in the research include the following: (1) fewer programs exist for high school students, (2) the percentage of students that are unsupervised during out-of-school time, and (3) students and parents do not want out-of-school time programs to be an extension of the regular school day, but offer a variety of activities based on input from students.

In closing, the review of literature supports the finding that out-of-school time programs are receiving more attention than ever before. Program accountability, evaluation methods, and proper interpretation of program findings are among the major issues in the field (Scott-Little, Hamann, & Jurs, 2002). The diversity of organizations housing out-of-school time programs, the population served, and desired program outcome make it difficult to identify a comprehensive set of key elements to quality programs. Therefore, it is a better practice for out-of-school time programs to develop their own quality standards rather than adopt a broad national standard which may not reflect the complexity and diversity represented in various OST programs (Tolman et al., 2002).

Although the majority of the research supported positive findings regarding after-school programs in relationship to student achievement and behavior, some studies reported negative results which included the United States Department of Education (2003b) that reported poor first year findings of the 21st Century Community Learning Centers programs. The study "reported minimal or no major impact on academic achievement and other program indicators" (U.S. Department of Education, 2003b, p. xii). This study further fueled the discussion regarding the efficacy of out-of-school time programs and the lack of quality evaluative studies by advocates and opponents of out-of-school time programs.

Research supportive of out-of-school time programs continue to acknowledge the critical role in closing the achievement gap specifically for economically disadvantaged youth by extending the school day and providing enrichment and academic support in a safe environment. However, the mark of success in out-of-time programs continues to be student retention and academic achievement. Finally, instilling the value of an education to all students is paramount to success in school and life.

Future research will address the variables related to the social characteristics and demographics of out-of-school time programs in relation to student achievement and the need for recognized OST quality elements that function across program types and geographical location. The cost of un-funded mandates as they relate to the quality standards will continue to be a challenge for program providers. Proposals to redirect OST funding which is part of the current education legislation may create a larger problem for providers if school districts are allowed to use the funding during the regular school day. Next, chapter three will provide the methodology, research questions and role of the researcher, data management and the summary.

Chapter Three

Methodology

The purpose of this intrinsic case study was to describe key elements of quality in out-ofschool time (OST) programs for school age youth as perceived by the participants. The study identified the standards for high quality out-of-school time programs that were linked to student achievement and positive outcomes for participants. A 21st Century Community Learning Center (21st CCLC) OST program for school age children in Southeast Arkansas was selected for this case study. The goal of the primary research questions was to identify quality components of the program studied as they were related to program outcomes measured by the experience of program administrators, staff, and participants.

The case study takes place in a Southeast Arkansas city of 9,146 people and with a fortymile radius population of 99,000 people (U.S. Census 2000, Economic Development Commission, 2010). According to the city's Economic Development Commission, the area is progressive and considered as the area's retail, recreational, and cultural center with evidence of twenty-five percent growth in retail sales over the past six years. The city is home to a state university and situated in a rural agricultural setting with a water port nearby and Union Pacific rail service. Households and families in the city are best described with U.S. Census (2000a) data.

The city represents 3,592 households of which sixty-four point five (64.5%) percent are family households with thirty-two point five (32.5%) percent family households with their own children under age eighteen present. Female headed households with no husband present represent eighteen point three (18.3%) percent of households and forty-two point five (42.5%)

percent of households represent married couple families. Social characteristics revealed seventyeight (78%) percent (4,130) of the population age 25 and over are high school graduates or have higher educational attainment; twenty-two point two (22.2%) percent (1,172) have a Bachelor's degree or higher. Two school districts serve the area; one named for the city and one for the county.

Marshall and Rossman (2006) described case study methodology as a means to focus on a program or an organization through the immersion of the researcher in the setting and with the research outcome dependent on the worldview of both the researcher and participants. This researcher applied Brantlinger's (1997) seven crucial assumptions of qualitative methodology as they related to the researcher's role in the case study approach selected for this study: (a) researcher viewed the nature of the research as critical with a political agenda to identify and promote quality OST programming; (b) researcher's position relative to participants was distant and objective as an observer; (c) direction of researcher's gaze was outward toward others, thereby externalizing the research problem; (d) purpose of the research was intended to be useful and informative to participants on site and other OST programs and policy makers; (e) intended audience of the study was the scholarly community, OST programs and policy makers; (f) the research was positioned politically with an agenda to enhance the quality standards of OST programs; and (g) researcher's exercise of agency was a part of OST policy practice.

Creswell (2007) identified three types of case studies including a single instrumental case study, collective or multiple case studies, and an intrinsic case study. This research takes the form of an intrinsic case study that focused on the case itself which for this study was the entire OST program at a middle school in Southeast Arkansas. Stake (1995) noted that intrinsic case studies were used to evaluate program elements. The focus of the case study under investigation was the program elements of quality and the impacts and perceptions of program quality for participants. The intrinsic case study resembled narrative research following a prescription of analytic procedures that required detailed description of the case study set within contextual boundaries of its physical surroundings.

Research Questions

The following two key research questions guided the study:

- 1. What are the elements of quality out-of-school time programs for school age youth as perceived by the participants?
- 2. What is the impact of quality as perceived by administrators, teachers, and students in out-of-school time programs?

Research design. According to Firebaugh (2008), the case study method best documents characteristics of program outcome and quality elements that may not become as evident with other research designs that do not allow for deep description of complexities. Qualitative research methods allow "thick description" (p. 26) and strategic data collection and can often extend or correct quantitative research findings. Coding textual data transforms interviews, field notes, and other documents into nominal variables which were, in essence, what statisticians mean by the phrase qualitative data (Bernard, 2000). The variables in this study revolved around quality elements of OST programs.

The research for this study was conducted in four phases: (1) semi-structured interviews with 21st CCLC personnel (administrators and site supervisor); (2) the collection of observational data (students/ students relate to staff; staff relating to parents); and (3) review of program

artifacts and documents from multiple sources. Each phase was intended to answer research questions and/or support overall findings. Findings were analyzed in phase four (4) through open coding of data for the purpose of identifying major themes.

The data collection process began with semi-structured interviews with participants, administrators and teachers. Marshall and Rossman (2006) define qualitative in-depth interviews as conversation-like with predetermined response categories to explore the participant's views while respecting how participant framed responses. It was important to identify the perspective of administrators and the site supervisor regarding quality programming. The goal of the interview was to establish the program's funding source(s) and role of parent and community partners in the program. Further, the interview design seeks to establish elements of quality in the program and how quality is organized in the program's design. The interview design also identified elements of quality in the program and explored whether improving academic achievement and/or meeting or improving students' developmental needs were among expected program outcomes. In this study, informal, semi-structured interviews were defined as interviews conducted in a relaxed, non-formal setting focused on gaining an overview of the program to be studied from program administrators and the site coordinator. Interview questions were listed in Table 2 as outlined:

Table 2

Sample Interview Questions

Questions for OST Personnel

- 1. What is the operational design of your OST- program?
- 2. How are elements of quality (best practices) organized in your program's design?
- 3. What are the funding sources for your program?
- 4. What are the desired outcomes of your program?
- 5. How do these outcomes manifest in your program?
- 6. Are your program outcomes measurable?
- 7. What are the ultimate goals of your program?
- 8. Is academic achievement a goal of your program?
- 9. Does your program meet developmental needs of students?
- 10. Who are your community partners?
- 11. How do you utilize your community partners?
- 12. How do you utilize parents as program partners?

The second phase of the conceptual design focused on observational data collection regarding the role teachers and program staff played during implementation and delivery of quality OST programming. Teachers were observed in the program setting while interacting with students and parents. The field note guide used for observation of teachers and program staff in Table 3 presented the theoretical framework for elements to quality OST programs developed by this researcher based on the Arkansas Governors' Task Force on Best Practices for Afterschool and Summer Programs (Beebe, 2008) and Maslow's (1954) Theory of Human Motivation. The perspective of teachers was particularly important to this researcher because they were on the front-line delivering quality programming to students.

Table 3

Field Note Guide

OST Program Observation

- 1. Observe how safety is designed in facilities and learning environments.
- 2. Observe ways students demonstrate the feeling of safety in program environment.
- 3. Observe instances of staff building trust with students.
- 4. Observe instances of staff building relationships with parents.
- 5. Characterize through observation staff interpersonal interactions with students.
- 6. Characterize through observation staff interpersonal interactions with parents.
- 7. Observe instances of program design or staff meeting developmental needs of students.
- 8. Characterize through observation evidence of student educational achievements.
- 9. Characterize through observation how staff recognizes student's developmental achievements.
- 10. Characterize how the program impacts students.

Observational data were collected by observing the program in action. Particularly, how students engaged in program activities and student's interaction with teachers and staff was of particular interest to this researcher. Students' interaction with teachers and program staff was monitored and documented through comprehensive field notes. Similarly, teachers and program staff interaction with parents were monitored and documented through the collection of field notes. Field notes were reviewed, documented, and transcribed into qualitative data ready for open-coding for the identification of major theme which formed the basis for the codebook and outlined the guiding principles. Field notes collection was used for observation of teachers and program staff in the OST program under investigation. The third phase of the research design played a supportive role.

In the third phase, the investigation of program artifacts and documents from multiple sources were used to support major themes appearing in interview and observational data. All data pertaining to the program in print and on the Internet were analyzed for congruency with qualitative data pertaining to the program operation. The operation of the program was revealed through interviews with the administrators, site coordinator, teachers, and staff and through observation of students and their interactions with the program and staff. Elements of quality found in the program operation compared to what the program promises to deliver in various documents were noted.

The following documents were analyzed: OST program web-based data for Arkansas 21st CLCC which was designed and maintained by MGT of America, Inc., including: (a) site profile; (b) student enrollment; (c) weekly activity log; (d) program reporting information; and (e) report data. Qualitative data compiled throughout the case study were open-coded to search for major themes and to identify elements of program quality.

In phase four, interview and observational data written from field notes were open-coded to identify themes present in the data and elements of program quality. A hierarchy was constructed to depict the findings of the elements of quality. Phase four also included an additional search of the literature to find a basis for comparison between the literature and the findings of this study.

The city's middle school was the host school of the OST program under investigation, and published the following mission statement:

[Our school] is committed to providing all students a challenging curriculum, quality instruction, and varied assessment aligned with Arkansas frameworks and standards. The school creates a learning environment that supports each student's intellectual, ethical, social, and physical development. The [school] faculty, staff, and administration actively

collaborate with family and community to help insure that students succeed in meeting goals and acquiring skills needed to be successful in high school.

The out-of-school time program purpose and goals were aligned with the mission statement of the school district. The OST program provided opportunities for academic remediation and enrichment activities that addressed the needs of the students. However, the priority was student achievement and performance on standardized test.

According to the National Center for Education Statistics' public school data for the 2008-2009 school years, the host middle school housed grades six through eight grade students and thirty-seven (37) classroom teachers with a student/teacher ratio of twelve point six (12.6). The Arkansas School Performance Report (2009) reported forty-seven percent of students qualified for free/reduced price meals. Table 4 highlighted the targeted school enrollment for 2008-2009.

Table 4

| Enrollment | Race/E | Ethnicity | Sex | | Subsidized Meals | | |
|---------------------------|----------|------------|--------|-----|------------------|-----|--|
| 6 th Grade 168 | 3 White | 320 | Male | 216 | Reduced Lunch | 26 | |
| 7 th Grade 152 | Black | 139 | Female | 249 | Free Lunch | 219 | |
| 8 th Grade 145 | i Hispan | nic 5 | | | | | |
| Total 465 | Asian/ | 1 | | | | | |
| | Pacific | : Islander | | | | | |

The Arkansas School Performance Report (2009) disclosed No Child Left Behind Adequate Yearly Progress (AYP) data and reported the school achieved standards in 2008/2009 and had met standards in mathematics, literacy, and attendance during 2009. The African-American subgroup met the Annual Measurable Objective (AMO) in mathematics: sixty-seven point one (67.1%) proficient status and seventy point seven (70.7%) percent proficient growth; met AMO in literacy: achieved sixty-three point seven (63.7%) percent proficient status and sixty-six point seven (66.7%) percent proficient growth. Economically disadvantaged students met AMO in mathematics: seventy-three point five (73.5%) percent proficient status and seventy-six point three (76.3%) proficient growth; met AMO in literacy: achieved sixty-five point four (65.4%) percent proficient status and sixty-seven point eight (67.8%) proficient growth.

The school report card for the host school revealed the majority of students were performing at the proficient and advanced level. Table 5 depicted performance levels for the host school students.

Table 5

| Population | Below | Basic | Proficient | Advanced | Proficient & Above |
|------------------|-------|-------|------------|----------|--------------------|
| | Basic | % | % | % | % |
| | % | | | | |
| Combined | 1.8 | 13.5 | 46.0 | 38.7 | 84.7 |
| African-American | 3.4 | 23.7 | 42.4 | 30.5 | 72.9 |
| Caucasian | 1.0 | 7.8 | 47.6 | 43.7 | 91.3 |
| Economic | 3.8 | 22.8 | 48.1 | 25.3 | 73.4 |
| Disadvantaged | | | | | |
| Female | | 8.0 | 44.3 | 47.7 | 92.0 |
| Male | 4.0 | 20.0 | 48.0 | 28.0 | 76.0 |

Annual Performance Report 2009

The purposeful selection of subjects for this study reflected the rational choice of the researcher to engage respondents with the deepest knowledge of program design, quality elements, and program outcomes. Participants were identified through researching the job descriptions and professional development information of administrative staff, teachers and

program staff to select the most knowledgeable interview subjects. The program itself was a pool of potential participants during the observation phase of this research project. Interactions of staff with students and staff with parents were the subject of observation, field notes, and later documentation of data and data coding.

Rationale for selecting school/program was two-fold: (a) the even distribution of demographic variables such as grade level, race, and socioeconomic status (whether students were from advantaged or disadvantaged backgrounds); (b) the program has been recognized for best practice. However a deeper investigation as to perceptions of program quality was necessary to determine the impact of quality on participants.

A broader inspection of the host school's performance report for 2009 revealed an increase in students performing below basic and basic in the African-American sub-population for seventh grade mathematics, seventh grade science, eighth grade literacy, and eighth grade mathematics. Figure 2 illustrates the percentage of African-American students in each performance level for each subtest.





Figure 3 represented a comparison between two sub-populations: African-Americans and economically disadvantaged middle school students. By the eighth grade, a dramatic increase was noted in poor mathematic performance. However, economically disadvantaged students were slightly behind African-American students in poorer performance indicators.



Figure 3. Comparison: African-American vs. economically disadvantaged middle school students

Researcher's Role

This case study examines a 21st Century Community Learning Center out-of-school time (OST) program which complements the professional background of the researcher. This researcher possesses over ten years' professional experience as a service provider and administrator of out-of-school time programs. Other applicable experience in the field included fund development demonstrated by successfully writing and administrating three 21st Century Community Learning Center (21st CCLC) grants for a primary and elementary school in Southwest Arkansas and co-chairing the Arkansas Governor's Task Force on Best Practices for After-school and Summer Programs.

This researcher is experienced in policy recommendations that support programs in a wide-range of settings, including, but not limited to school districts and community and faith-

based organizations, by providing opportunities for children and youth to engage in quality outof-school time and summer programs across the state of Arkansas. As a professional providing service for school age children, the benefits of OST programs were evident and can make a difference by providing quality, accessible out-of-school time programs that keep children safe, help working parents, and improve academic performance.

Historical studies also provided an opportunity for the researcher to analyze and evaluate a vast amount of research. Identifying important elements in the literature relating to significance for this case study continued to be a challenge throughout the study. As a result of an ongoing review of best practices and quality standards on the state and national level, this study provided an opportunity for the researcher to examine the standards currently in place for 21st CLCC out-of-school time programs in relationship to the Arkansas State Standards.

Data Management

This case study was conducted during the 2010-2011 school year. During the summer of 2010, the researcher began literature search and review. The fall of 2010, the researcher began documentation collection with approval of the school district and Institutional Review Board (IRB). The spring of 2011 interviews were conducted, data analyzed, and the findings will be presented upon committee approval.

All data gathered in this case study were regarded as confidential. All documents were secured in a protected setting in an effort to maintain confidentiality and anonymity. Supporting documents were date stamped, coded by categories, and filed during the data analysis process. Program artifacts and documents were used to validate the study. Artifacts and documents were procured by submitting written request to program administrators and the site coordinator.

The data management process streamlined the data collection and analysis process. Data were analyzed as collected and sorted into themes through open-coding facilitated by the use of a code book developed from interview questions. Interviews were coded from interview sheets and field notes were coded using the field note guide. Descriptive coding was used to analyze and summarize the primary topics pertinent to the study. Data were assembled and electronically stored by emerging theme for collective analysis as the study progressed.

Data collection. Marshall and Rossman (2006) presented positionality guidelines as a method to disclose researcher participation while conducting observations. This researcher established the following planned extent of participation while at the research site: (a) the researcher was a passive participant in program activities; (b) the study's purpose was not fully disclosed to people in the setting; (c) the researcher was presented as a program observer; (d) the researcher's participation focused on observation during the collection period; (e) ethical dilemmas pertaining to quality program objective disclosure was managed through Institutional Review Board (IRB); and (f) collection of observational data answered the research questions through identifying elements of quality as perceived by participants and recording participant's perceived impact of quality programs.

The following process was used to conduct interviews and observations. An observational protocol based on Creswell (2009) outlined how observational data were collected. Field notes were gathered by conducting observation as an observer. Field notes were divided into two columns: (1) the right hand side column was reserved for descriptive notes including character sketches of participants, dialogue reconstruction, description of the setting, and accounts of program activities and events; (2) the left hand side column was reserved for reflective notes which may contain the researcher's thought responses regarding what was being observed, and demographic information including time, place, and date in which the observation occurred.

Creswell (2009) recommended adoption of an interview protocol—this study used the protocol to ask interview questions and record responses: (a) interviews were documented on a prepared interview sheet with a heading disclosing date, place, interviewer, and interviewee; (b) interview procedures were standardized with interviewer instructions included on each sheet; (c) interview questions were supplemented with opening and closing questions: (ice breaker question and a follow up question); (d) Probes were inserted for each question in case participant elaboration or explanation required prompting; (e) Space was provided for interviewer to record responses; and (f) a thank you statement appeared at the end of the interview form to acknowledge participant's investment of time.

The concept of saturation was used as a research strategy in data collection to ensure credibility of the study. Phase one and two of the research study was on-going until data saturation occurred. Data were analyzed as phase one and two progressed for the purpose of discovering patterns that related specifically to the research questions. Interviews and observations were discontinued after consistent patterns in the data were documented and no new patterns or themes emerged from the data. Triangulation analysis to validate emerging themes in the data were used and once completed signaled to the researcher that data saturation had occurred. When data saturation occurred, the research study progressed toward completion of findings in Chapter Four.

Data analysis. The goal of data analysis was to arrive at themes relating to program quality. The data analysis process included open-coding. Open coding involved reading the

transcripts, identifying, labeling and categorizing related and recurring themes. Open coding is appropriate for this case study because the process included descriptive note taking from observation, interviews and collecting pertinent documents that were coded and labeled.

Creswell (2009) recognized qualitative data analysis as a progression through interrelated steps. This study adopted an interactive approach to data analysis: (1) data were organized and prepared for analysis by transcribing interviews, documenting field notes, and sorting data types; (2) data were assimilated by the researcher by reading thoroughly and reflecting on overall meaning and making comprehensive notes; (3) items will be coded into general category themes specified in a booklet relating to elements of program quality and codes that were surprising, unusual, or that represented a theoretical perspective in the research; (4) the coding process included a description of the setting, participants, and themes for analysis; (5) description and themes were represented in narrative form; and (6) an interpretation of data produced a wider meaning including lessons learned, comparison of findings in literature and theories, and future research questions for investigation.

The distribution of themes listed in Table 6 reveals a consistency of comments of all staff interviewed regarding perceptions of students. The participant's code (numbers) at the top of the table identifies each participant. All interviews except one certified teacher made references to the program. Interview content form all interviews except one certified teacher made references to grades. Interviews demonstrated recognition of the program focus through references of student achievement, high-stakes testing, and homework. Interviews reveal an understanding of program strengths, funding, and program sustainability and outcomes.

Table 6

Distribution of Categories & Themes

| Participant Codes | 0101 | 0102 | 0103 | 0104 | 0105 | 0106 | 0107 | 0108 |
|---------------------------------|------|------|------|------|------|------|------|------|
| Categories | | | | | | | | |
| Students | Х | Х | Х | Х | Х | Х | Х | Х |
| Relationships | | | Х | Х | Х | | | |
| Student Achievement | | Х | Х | | | Х | Х | Х |
| Program Sustainability/Outcomes | | Х | Х | | Х | Х | | |
| Themes | | | | | | | | |
| Safety | | | Х | Х | | Х | | |
| Structure | | Х | | Х | Х | | | |
| Program | Х | Х | Х | Х | Х | Х | Х | |
| Caring | | Х | | | Х | | | Х |
| Strong | Х | | | Х | Х | Х | | |
| Expectations | | | | | Х | Х | Х | |
| Grades | Х | Х | Х | | Х | Х | Х | Х |
| Homework | | Х | Х | | Х | Х | | Х |
| Choices | Х | | Х | | | | | |
| Funding | | | Х | Х | | Х | Х | |
| High-Stakes Testing | | | Х | Х | | Х | Х | Х |
| Partners | | | | Х | | | Х | Х |

Trustworthiness and credibility. The credibility of findings was addressed by ensuring findings were based on reliable information from credible respondents and informants, prolonged and persistent engagement, triangulation, peer debriefing, member checks, and the establishment of an audit trail. Undue researcher bias was avoided by making data available for peer review by three faculty members in the Educational Leadership Program from the local university early in the study and continually as the study progressed.

Prolonged engagement. The data collection process was completed over a period of five months. The researcher has over a decade of professional expertise in OST programs which allowed for thorough data collection in a compressed time period. Findings were validated through the various phases of the research design.

Persistent engagement. Persistent engagement was accomplished through the reinterview process utilized to check the data for inconsistencies. Data inconsistencies were verified through supplemental interviews and member checks. The respondent's perspective of the program, knowledge of best practices and quality elements, professional development, program design, proposed program outcomes and actual program outcomes were considered in the evaluation of inconsistencies in responses to interview questions.

Triangulation. The primary source of data utilized in this study were interviews. Conversational and semi-structured interviews were conducted at the host school site. Program artifacts and documents from the host school, state department, briefs, and the school district's website were collected and utilized to provide additional data. Through the use of multiple sources of data, the researcher was able to complete the triangulation process for this study which increases the credibility and validity of the results.

Member checks. Interviews and other qualitative data were documented and shared confidentially with key respondents for their reaction to the interview. These respondents were invited to expand, clarify, or correct responses to interview questions. In some cases, follow-up interviews were completed to address or clarify potentially incongruent data before member checks were completed.

Audit trail. An audit trail was established as a means to confirm data. Data were securely stored electronically and on a computer USB storage device. All aspects of data for this study were secured including (a) recordings of interviews; (b) interview transcripts; (c) interview questions; (d) collected documents and artifacts; (e) field notes; (f) results of data analysis; and (g) results of document analysis.
Summary

.

The qualitative research study provided an extensive and comprehensive process of data collection and analysis. Triangulation methods and qualitative data coding techniques to identify major themes were the foundation for the interpretation of findings. The interview questions reflected the theoretical framework for quality OST programs and guided the research in an effort to answer the research questions. The research design of this study, data collection and analyzing techniques, followed by findings, discussion and conclusions of this study were executed to clearly answer the research questions as further outlined in Chapter 4. Chapter 4 presents the data analysis and management process, research questions, elements of quality, description and distribution of themes and categories and summary of data.

Chapter Four

Data Presentation and Analysis

Introduction

The purpose of this intrinsic case study was to describe the key elements of quality in out-of-school time programs for school age youth as perceived by the participants. The goal of the study was the identification of standards for high quality out-of-school time programs that are linked to student achievement and positive outcomes for participants in a 21st Century Community Learning Center. The intent of the federal No Child Left Behind Act of 2001, (NCLB, 2002) is for all students to become proficient by 2013-2014 on standardized tests by holding schools accountable for student performance including subgroups within a school district.

This chapter includes a review and discussion of key findings correlated to the research questions. The researcher interviewed key participants of the target group for the study. Program artifacts, documents, personnel interviews, and observational data were used to address the research questions.

The following research questions were addressed:

Research Question:

- 1. What are the elements of quality out-of-school time programs for school age youth as perceived by the participants?
- 2. What is the impact of quality as perceived by administrators, teachers, and students in out-of-school time programs?

Description of Participants

The participants selected for this case study included middle school students, site coordinator, administrator, certified teachers and college students. Participants self-reported information regarding qualifications, background and experience including national board certification. This information was documented and maintained in a separate file to maintain confidentiality. Eight certified teachers, a site coordinator, a program administrator, and two college students were interviewed for the study. Ten middle school students were interviewed; however, due to limited content responses from the students, responses were summarized. Of the eight certified teachers, one teacher reported being national board certified and another teacher reported enrollment in the doctoral program at the local university. The site coordinator reported eleven years of employment in the out-of-school time program.

Additional information obtained through school records and personnel included the student enrollment numbers that were listed as 2,082. There were 480 students enrolled in the middle school. The composition of the student body was 35% Black and 65% White, less than 1% other. The makeup of the faculty in the out-of-school time program was predominantly white and black females, the males present were college students. Table 7 listed below provided a description of the program participants which included faculty and students of the host school.

Table 7

Description of Participants: Faculty (F) and Students (S)

| Sex | | Participants | Race | |
|---------|-----|--------------|-------|--|
| Females | (F) | 7 | White | |
| Females | (F) | 3 | Black | |
| Male | (F) | 1 | White | |
| Males | (F) | 1 | Black | |
| Females | (S) | 2 | White | |
| Females | (S) | 3 | Black | |
| Males | (S) | 1 | White | |
| Males | (S) | 4 | Black | |

Data Management

The data collection techniques included the written approval of the Superintendent to conduct the study in the school district. Parental consent forms for students and permission forms for personnel were completed and approved in the entrance phase of study. Initial visits were conducted to become familiar with the school district, personnel, students and the operation of the out-of-school time program. Access to documents including demographics, policies and procedures pertaining to the out-of-school time program, test scores, and the Arkansas Consolidated School Improvement Plan (ACSIP) was granted to the researcher by school personnel. Strategic locations were identified for conducting the interviews and observations.

The researcher manually completed the data analysis process. Data were analyzed as collected and sorted into themes through open-coding facilitated by the utilization of the code book and the field note guide. Interviews and field notes were coded from interview sheets and field note guide.

Descriptive coding was used to analyze and summarize the primary topics pertinent to the study. The process included reading the interviews and field notes multiple times. The next

phase included identification of patterns, major and recurring themes, and examination of documents that identified elements of quality in programming and outcomes. Data were coded into general category themes specified in the code book relating to elements of program quality. Supporting documents were date stamped, coded and filed during the data analysis process. The participants were listed and identified by the coding system. Audit trail notations were used for certified teachers which were identified as CT followed by a numeral code of 01, 02, and 03. College students were identified as CS and followed the same pattern with numerical codes of 09 and 010. Direct quotes from participants are also included in this chapter and are identified by numerals and letters. The interview questions are identified by numbers and letters that indicate the response from the interviews. Table 8 provides an illustration of the register of audit trail notations for participants.

Table 8

| Notation | Participants | Questions | Codes |
|----------|------------------|-----------|-------|
| PM | Project Manager | District | 0104 |
| SC | Site Coordinator | MMS | 0103 |
| СТ | Teacher | MMS | 0101 |
| СТ | Teacher | MMS | 0102 |
| СТ | Teacher | MMS | 0105 |
| LB | Librarian | MMS | 0107 |
| СТ | Teacher | MMS | 0106 |
| СТ | Teacher | MMS | 0108 |
| CS | College Student | MMS | 0109 |
| CS | College Student | MMS | 0110 |

Audit Trail Notations

The interviews were recorded using a digital recorder and transcribed electronically.

Triangulation analysis was used to validate and complete emerging themes in the data until data

saturation occurred. According to Merriman (2009), triangulation involves the use of multiple sources of data that includes cross-checking data that were collected through observations at different times and places or interview data collected from different people with various perspectives or from follow-up interviews.

Interviews and other qualitative data were shared confidentially with key participants for their responses to the interview. Participants were invited to review, clarify or correct responses to interview questions. To protect the anonymity of participants, parenthesis () were used by the researcher.

Research Question One

The first research question was addressed through a semi-structured interview process that included conversational and open-ended interviews with the site coordinator and middle school teachers at the selected location. The initial conversational interview was conducted with the administrator. The administrator provided general program information regarding the out-ofschool time program that included the hours of operation and the quality of the certified staff and college students that worked with the students. Additional information was provided by the Project Manager in a conversational and semi-structured interview format. The Project Manager described the out-of-school time program as an extended program focused on student achievement.

The students needing help in literacy to meet the Accelerated Reading Program goals were referred to the out-of-school time program and placed on an academic improvement plan. However, some students attended for homework completion and to participate in the enrichment activities which included Archery and Fitness. The overall goal of the interviews was to identify the perspective of the personnel related to quality programming in the out-of-school time program.

Elements of quality. The following categories define the elements of quality identified by participants in the out-of-school time programs. The categories were students (participants), student achievement, program sustainability/outcomes, and relationships. The emerging themes listed under the category of students included safety, structure and program. The second category was student achievement and the emerging themes were grades, homework, high stake testing and choices.

The third category identified was program sustainability/outcomes and the emerging themes were funding, support and partners. Relationship was the fourth category and the themes included caring, strong and expectations. There were a number of similarities in the program operation and sequence of activities correlated to the emerging themes. Listed in Table 9 are the categories and themes related to program quality in a 21st Century out-of-school time program.

Table 9

Elements of Quality

| Categories | Participants | Student Achievement | Program Sustainability/Outcomes | Relationships |
|------------|--------------|------------------------|------------------------------------|---------------|
| Themes | Safety | Grades | Funding | Caring |
| | Structure | Homework | Support | Strong |
| | Program | Choices | Partners | Expectations |
| | Program | Choices | Partners | Expect |

Description of Categories and Themes

The themes emerging from the data were linked by categories as determined by the data analysis process. Three emerging themes were listed under each category. Interview content regarding student achievement was found to be the richest category with related themes of grades and homework. Interviews related shared themes regarding the program for the participants' category. Key elements of quality identified in the relationship category were demonstrated by staff and student interactions and during interviews process. Program sustainability was a major concern for staff and the future of the program. Key elements described by staff included the need for financial support, resources and community partners.

Student Participants

The first major category related to program quality was student participation in the outof-school time program. The participants included a diverse population from various socioeconomic backgrounds that were enrolled in the program because of basic or below basic scores on the benchmark examination. Students were provided an array of enrichment opportunities within a safe and structured environment. Similar activities were not available to students outside this program. One teacher's description of why students attended the out-ofschool time program:

Several factors may influence why students attended the out-of-school time program. One major factor was the benchmark scores. If students scored basic or below basic on the benchmark, they were assigned either 30 or 45 days in the out-of-school time program. If students scored basic or below basic in Math, the students were assigned 30 days. If students score basic or below basic in Math and English, they are assigned 45 days. So that will be one factor and then too, we have a lot of kids that like to come for Encore classes outside the day school which is a chance for them to do something different. For example, Archery and Project Alert, a self-esteem building program, social type things. There are many different reasons, quite a few come because they have to, but we do have students that do not have to attend, that do come. (CT-010l)

The students enjoyed a variety of activities offered in the out-of-school time program. Some of the activities include Project Alert which is an enrichment activity. The Writing Workshop was offered to students and provided an opportunity for students to integrate technology through the use of flash drives, digital cameras, and lap top computers. Students were introduced to a variety of science projects and they were encouraged to do experiments. Students were taught how to set up an experiment and write reports. Through the science activities, students were introduced to new concepts that stimulate the imagination. Online activities were also included as a part of science and the integration of technology.

Safety. The first theme identified by administrators and teachers under the participant's category was the safety factors related to the out-of-school time program. According to the

district's handbook, the middle school campus is a closed campus which encompasses the out-ofschool time program. Students must remain on campus and leave only with permission of the administrator and the parents. Visitors must report to the office. Students are monitored by video and a camera system. Violations of school regulations and policies may result in detention or suspension depending on the severity of the infraction which is applicable to out-of-school time program participants. The importance of providing a safe environment and security for students and faculty was evidenced by the school policies and the freedom of movement of the students and staff. Nutritious snacks were provided for each student and physical activities through a Fitness Program incorporated into the enrichment activities to address the health needs of students.

Several teachers reported the out-of school time program followed the same policies and procedures as the regular school day in regard to safety. The Student/Parent Handbook outlined the responsibility of the school district:

maintain discipline, protect the safety, security, and welfare of its students, staff, and visitors while at the same time safeguarding facilities, vehicles, and equipment. Video/audio surveillance cameras are used in school facilities/grounds and school buses. Students are held responsible for violations of school discipline rules caught by the cameras. (p. 60)

Also, an emergency drill was conducted monthly which included fire drills and tornado drills during the months of September, October, January, and February. An emergency evacuation drill was conducted twice a year for bus riders. Other emergency drills included lock-downs, acts of terrorism and chemical spills. The emergency drill and procedure was in compliance with the school district policies.

The safety elements of the out-of-school program related to quality were consistently identified by the teachers in multiple interviews. The teachers linked the school policies and procedures to the out-of-school time program design. One teacher's description of quality related to safety included the criminal background checks required by the school district for staff and a system of checking students in and out of the programs by teachers and parents. Further comments by a teacher included her opinion that the "out-of-school time program was a very safe place for students and the quality of the program was good." (CT-0101) Several teachers indicated they wanted the program to be safe and make a difference. Also, school personnel related even though it was an out-school time program, they were in compliance with program standards and followed the school district policies and procedures.

Since the classes and enrichment activities were located on the middle school campus, the building security system was operational during the out-of-school time program. The custodial staff provided additional assistance with building supervision after students leave for the regular day. There was a sign in/out system in place for the parents to check students out through the office. The students riding the bus were checked out by the staff before leaving the program.

Program structure. The overall structure and purpose of the out-of-school time program as it related to program quality was defined within the district's Arkansas Consolidated School Improvement Plan (ACSIP). The plan required by the Department of Education identified the priorities for the school district. The school district linked the academic goals for the out-of-school time programs in the plan. According to the plan, the middle school students must meet the proficient level of performance in literacy and mathematics by the end of the 2013-2014 school years. The school must also make adequate yearly progress (AYP) toward meeting that goal. Failure to meet these standards resulted in the school district being placed on the School Improvement list if less than 95 percent of eligible students were tested or the school does not meet the secondary indicator (school attendance).

According to the site coordinator, the program was originally designed with input from the team which included the principal, staff, teachers, and counselors. Surveys were also used to determine the needs of the students. The need and desires of the students and parents were meshed with the goals of teachers and the quality assessment goals. The goals of the 21st Century Community Learning Center funding were linked and addressed in the program objectives.

Further information on the structure and program design related to quality was provided by an administrator. According to the teaching staff, the site coordinator provided the directives for the program and maintained records of student progress while serving as a resource person to the out-of-school time staff. One teacher described the structure of the program:

We can do things in afterschool that we can't do in day school. They can get up and run around outside with 8 to 10 students that you can't do with 20-25 kids in a classroom. The strategy that we use, we target students that are below proficient, we mandate students that are below proficient, it is part of our ASCIP, 30 days if they score basic, 45 days, if they score below basic. We see after-school as a tool to help them improve their performance on the benchmark and other high stake exams. (SC-0103)

The out-of-school time program is supervised by certified teachers assigned a 30 minute block for a particular subject area. The enrichment activities were coordinated by the certified teachers and college students.

Out-of-school time program. According to the teachers, the out-of-school time program environment provided the support and resources the students needed to be successful in the regular day classroom. Assistance was available for homework completion and supplies were provided through the library to help student's complete special projects. The Behavior Intervention teacher provided extra support and supervision for teachers during the out-of-school time program by working one-on-one with students that were challenging and may have special needs.

The Librarian was on duty to assist students with reading materials and the computers were available for students. The students had choices and options in the out-of-school program that provided an opportunity for students to give their ideas and suggestions to the teachers regarding the type of enrichment activities they would like to see incorporated into the program.

The school district's facility provided the space, equipment and resources needed to operate an effective and quality out-of-school time program. The program environment allowed freedom of movement for students and staff without safety concerns due to the security system and the monitoring of the students by personnel.

Additional information obtained from the *Parent/Student Handbook* described the out-of school time program as a program that started the first week in September and ends the last week in April. The out-of-school program included a remediation and enrichment component that is provided Monday-Thursday. Students were transported by parents or the school bus.

Several teachers described the composition of the out-of-school time program as four thirty- minute blocks Monday through Thursday from 3:20 p.m. – 5:20 p.m. The classes included the core subjects and the enrichment classes. The classes were Math & Science Workshops, Writers & Nutrition Workshop, Library, Homework, Archery and Physical Education. Behavior Intervention was provided by the Special Education teacher for students with challenging behaviors or students that needed help working through personal problems that were related to school or home. Teachers also related that the block schedule allowed students that needed extra help with a certain subject or extra homework time the flexibility to remain in one area without rotating to another class. The students were also allowed a choice of enrichment activities.

Student Achievement

The second major category related to program quality identified by the administrative and program personnel was student achievement. The major goal of the out-of-school time program was to improve student performance on the local, state and national assessments. One teacher's assessment of the out-of-school time program as related to student achievement:

I think it started with the regular school day where everything is focused on student performance and students meeting the AR goals. Students wanting to participate in the out-of-school time program may choose to do so. However, it is placed on the students' academic improvement plan. The research shows even enrichment activities leads to academic performance. (PM-0104)

The site coordinator acknowledged "every student is not advanced and not every child is an A student in every subject area. However, by helping every child meet his or her potential, it will

help the school stay in the forefront of success." The 21st Century Community Learning Center OST program may provide additional resources to help students that may not be available during the regular school day.

Student grades/test scores. The first theme under the category of student achievement was identified as grades/test scores. Several teachers identified the strategies used in the out-of-school program to target students. According to the teachers:

The students are targeted that are below proficient, we mandate students that are below proficient, it is part of our AIP, 30 days if they score basic, 45 days, if they score below basic. We see after-school as a tool to help them improve their performance on the benchmark and other high stake exams. As a teacher, the focus is getting the students at this age to buy in that it is their job, their responsibility to learn and we will do anything we can to help them. (SC-0103)

Several teachers related that positive high-stake testing outcome as a core program goal during the interview process. One teacher opinion of the benchmark exams:

The students come to the out-of-school time program because of the benchmark examination. If students do not do well on the benchmark, they are mandated to come. For some students, it is homework; the parents want them to get their homework

completed before they get home because they may not be able to help them. (CT-0105) Another teacher stated, "most of the students attend for benchmark remediation; they are required to attend a certain number of days if they didn't score proficient or advance on their benchmark examination." (CT-0108) Additional interviews provided some rich descriptions of activities that were centered on high-stake testing in the out-school time program: Even though we are a block schedule, the students get Math every day; it is double blocked 90 minutes. We use to have seven periods during the regular day which really turned us around. We use ARA reading everyday to support literacy especially the reading component. We do reading and math interventions. We pull out the bubble students and target them. We have through the curriculum developed tons of writing activities. Even in Health and P.E., everyone has tried to support the writing. The ASCIP plan is tied to the out-of-school time program. If a child is below proficient, they are assigned 30 days for basic and 45 days below basic. If they don't come, they are mandated to summer school. (SC-0103)

The teachers further discussed the importance of helping those students that scored basic or below basic with various teaching strategies to become advanced or proficient. A certified teacher description of the program outcomes:

I think our program outcomes are measurable. One thing I know is our site coordinator looked at test scores before the students started afterschool for improvement or growth in afterschool. We track attendance, and we let the students know every nine weeks how many days they have attended, you can see the difference in their day school scores and attitude. (CT-0106)

Multiple teachers discussed the Benchmark examination during the interview process and the impact of the test scores on the school district related to meeting adequate yearly progress (AYP). If a school fails to meet AYP for two consecutive years, the school is listed as in need of improvement Year 1 and must offer public school choice according to the No Child Left Behind Act of 2001 (2002).

Homework assignments. The second theme under the category of student achievement is home work. The out-of-school time program provided an opportunity for students to complete their homework and receive additional help and support if needed from certified teachers. The Librarian was on duty until 5:20 p.m. to assist students with materials and the use of equipment for special projects. Computers were also available for student use. College students were available to provide tutoring and extra support to students that may be struggling in certain subject areas. According to the staff, the goal was to help students complete their homework assignments, stay on track in the regular classroom and improve their performance on the benchmark and other high stake exams.

Imbedded choices. The third theme under the category of student achievement was choices. According to the teachers, choices are imbedded in the out-of-school time program through the enrichment activities that are provided for students. The activities may include the writing program, Archery, Nutrition, and Physical Fitness. The programs were mentioned by several teachers. These programs are designed to be Monday through Thursday programs, three hours a day, snack, and transportation if needed. Students are divided by grade levels. They have choices and options which was one the quality assessment goals to give the student choices; therefore, there are embedded choice programs. All students have a structured rotation program. If the students elect not to participate in one of the optional programs; they are scheduled for a 30 minute block that included a variety of classes. Usually, the students are all together. Overall, the objective was to rotate every 30 minutes to give every student the option and choice to participate in a variety of activities.

Program Sustainability/Outcomes

The third major category related to program quality identified by the program staff and classified as critical to the overall success of the out-of-school program was program sustainability and measurable outcomes. The majority of the teachers referred to the Site Coordinator when questions were asked relating to program sustainability. The Site Coordinator described the district's position regarding sustaining the out-of-school time program when the 21st Century Community Learning Center funding end as having no way to sustain the program. The Site Coordinator further indicated the parents cannot afford to pay and there was no industry in the city, however, the majority of the parents were employed. The Site Coordinator felt it was unrealistic to think that the district determined the benefit of the out-of-school time program in relationship to student achievement was significant, the district agreed to pick up the transportation cost.

Academics were driving the out-of-school time program with the overall goal to improve test scores. The program outcome related by one teacher was stated as follows in response to the interview question: 'I hope students were able to make better scores on the benchmark examination; I hope they become better-rounded students. I just hope they have gained, that the biggest things, they are better off at the end, than when they started." (CT-0101) Teachers were held accountable for the performance of the students on the benchmark exams and expressed the critical role OST played in helping students improve test scores and become well-round students. The improvement in test scores and letter grades in the regular classroom continued to provide evidence of the value and contribution of the out-school-time program. **Federally funded.** The first theme listed under the category of program sustainability/outcomes was funding. The major funding source for the out-of-school time program was the 21st Century Community Learning Center funds which supported the school and community based programs. The funding provided the budget for the school district to employ certified teachers and highly qualified paraprofessionals to work in the program. Supplies, equipment and some of the food cost were covered with the 21st Century Community Learning Center funding.

The school district provided in-kind contributions consisting of accounting, janitorial, building space, utilities, equipment, playground, library and transportation. Since it is a school based out-of-school time program, the school district provided a safe environment that included a security system along with support personnel to monitor the building. Technology and food service support was also provided through the district.

Support/partners. The second and third themes listed under program sustainability/ outcomes were support and partners. The school district is a key partner in providing an out-ofschool time program for struggling students. The support of the school district manifests itself through qualified personnel and the overall administrative support of the district. The project manager's description of the involvement of the district: "The buy in of the administration was as strong as indicated by the manager. Even though we had different principals on board, the principal makes all the difference in the world." (PM-0104) The support of all the staff was noted as critical. Multiple teachers mentioned community service opportunities and volunteers from the football team and Greek organizations. Parent volunteers were also recruited to read with students, particularly the reluctant or struggling readers. The teachers related the students realized they can learn from reading and enjoy reading books, magazine and other printed materials.

The partnerships as indicated by teachers included a community outreach with the focus of getting the community involved in the enrichment activities of the out-of-school time program. A successful collaboration of African-American churches supported by 21st Century funding has helped by working with the churches and parents to understand what the students are facing on high stake testing and the impact of test scores on the academic success of the students. Finally, the impact of quality as perceived by administrators, teachers, and students in the out-of-school time program was related to program sustainability and academic achievement. The main goal related by multiple teachers was to improve the student performance on state, national and local assessments.

Relationships

The fourth major category related to program quality identified by the out-of-school staff was the importance of positive relationships with the students. According to the administrative support staff, out-of-school time staff was able to build unique relationships with the middle school students. The Site Coordinator expressed a concern regarding the composition of the staff being primarily white females. Therefore, the use of the university football players enabled communication with the male students and encouraged completion of homework in a manner different from the traditional staff. "The student came in today and said, 'I know my paper is right,' a football player sat beside me for 45 minutes and if I made a mistake all he said was, 'no man that's not right." (SC-0103)

The homework completion time block provided an opportunity for the students to receive

assistance and the use of computers and other materials which may not be available in the home. The majority of the parents are working according to staff and really appreciated and requested the completion of homework assignments.

Caring. The first theme under the category of relationship was caring. Staff demonstrated positive interactions and respect for the students while engaging in activities. One teacher's description of the interaction between staff and students:

The buy in from staff in getting the students to realize that we aren't here just because we draw a paycheck was huge. We are here because we care about you and we want you to do well, so the follow-up, the mentoring our staff gives the students. I will help you study for the test, but you must tell me how you did on the test. I am invested in you, how did you do? There are tangibles like stickers, suckers, or mom coming in saying this child is failing in everything, he is sitting there faking you out. Kids put more time into faking than they do actual work. We work with our students on positive intervention, teaching them how to study, how to organize their binders or making positive academic interventions. (SC-0103)

The instructional activities were age-appropriate and addressed the individual needs of students. positive interactions with staff. The group size was small which enhanced the amount of time teachers spent with each student. The relationship between the staff and students included not only a high level of respect, but strong evidence of trust and a sincere desire to see that the students were successful in school.

Strong program. The second theme under the category of relationship was strong program. According to the staff, the strength of the program centered on being student focused and the importance of the students recognizing the staff cared about their welfare and the success

of the program. The staff formed relationships with students through positive roles and recruited college students they could relate to which enhanced the program. Several teachers indicated the need for the out-of-school time program to make a difference and be a strong program. One OST teacher further described her program method:

The program is not all about academics, it includes some social things, life skills and things that are motivating to the students. The students are encouraged to think long term instead of short term. Middle school students are so short term; it is difficult to get the students to see the bigger picture. It helps to not constantly pound academic, but try to incorporate life skills and show the students how things relate outside the school building. (CT-0101)

Other teachers discussed the importance of providing a well-rounded program which included academics and enrichment, however, the program must be more than just academics to keep the students involved and engaged.

High expectations. The third theme under the category of relationship was expectation. The importance of helping students to build their self-esteem was repeated by several teachers. Examples were given by teachers of teaching the students the concept of "paying it forward" which meant after the students received help in the out-of-school program, they are encouraged to help someone else become a better student. A career teacher described her classroom incentive program for students as follows:

Mostly, it was just praise; they don't get enough attention, giving the students positive praise, I give a lot of little prizes; I had some little key chains, they love those. I had some little toys I had saved from cereal boxes, I am a pack rat and I save everything. I

had some little race cars, the boys loved them. They like being recognized. They want to read, they volunteer to read. When we are writing about our experiences, they wanted to read. I was really impressed with one group of boys they were so unruly, I thought, " oh my gosh," but they wrote the best stories and they all wanted to read them in front of the class which is improving their self esteem. That is the most important thing when working with the students to build their self-esteem. (CT-0102)

Teachers used various forms of incentives to recognize students and their accomplishment in the program.

Research Question Two

The second phase of the research design focused on observational data collection regarding the role of the teachers and program staff in the delivery of the program elements. Teachers were observed in the program setting while interacting with students and parents. The focus of the observation process was centered on the purpose of the study, theoretical framework and research questions. The researcher observed specific events and behaviors that occurred in the classroom setting, library, gym, and hallways that involved out-of-school time program students and staff. Students' interaction with teachers and program staff were monitored and documented through comprehensive field notes.

Observation notes were recorded on a code sheet that included the frequency of specific events and the physical setting or location. The interactions and conversation including nonverbal behavior between the staff and the students were also observed and recorded. "Observation is the best technique to use when an activity, event or situation can be observed firsthand," according to Merriman (2009, p. 117). The researcher's role was strictly as an observer from a distance to avoid any intrusive contact with the students or staff during planned and unplanned activities.

The participants entered the out-of-school time program location in an orderly manner and sat at various tables after the school bell rang for dismissal of the regular school day. Conversations between respondents were in a low tone of voice and smiles were exchanged as respondents removed their books from the backpacks and began working on assignments. Some students started reading books and others asked for assistance. The staff provided requested materials to the students while engaging in a conversation with a smile and playful interaction with the participants. Respondents moved to and from different classrooms in a 30 minute block rotation and some respondent remained in the initial location. The following field note from observation in a writing class:

The students (10) completed a writing assignment. The students used writing prompts and the teacher provided prompts as she walked around the classroom. The majority of the students present were males. The classroom bulletin boards contained information about writing skills which the teacher directed the students to read and follow the instructions. There was a mobile lab in the classroom the teacher used to assist students in various writing activities. The mobile lab contained lap top computers the students could use or if they were more comfortable writing with a pen, it was their choice. However, some of the writing activities required research for background information. Upon completion of the writing assignments, the students were required to print the assignment out or turn in a handwritten copy to the teacher. Overall, the observation time periods provided an opportunity for the researcher to view the outof-school time program firsthand as well as staff. Observations were recorded which included the classroom setting, participants' role, activities and interactions. Conversations were summarized and paraphrased to describe the content of the conversation. Codes were utilized to record events, behaviors and participants.

The review of program documents related to the out-of-school time program were categorized to include records, test documents, test scores, ACSIP Plan, Policies & Procedures, Professional Development & Training records, 21st Century Community Learning Center grant application, OST student rosters/schedules, School Report Card and the OST Handbook.

Summary of Data

The researcher presented the major categories and themes that were identified through the data management process in Chapter 4. The major categories developed from the interviews were: students, relationships, and student achievement and program sustainability/outcomes. The themes under the four major categories were outlined in Table 6. The data management process was organized and coded according to the two research questions. Chapter Five will present the introduction, summary, research questions, interpretation of data, field and program recommendations, recommendations for future research and the conclusion.

Chapter Five

Conclusion and Recommendations

Introduction

The purpose of this intrinsic case study was to describe the key elements of quality in out-of-school time programs for school age youth as perceived by the participants. The process included interviews, observational data, collection of documents and program artifacts. This chapter will include findings related to the data, conclusions and recommendations for current and future studies that have implications for educators and policy makers.

Summary

Improving student achievement and promoting the growth and development of youth are two of the major goals of out-of-school time (OST) programs. However, the growing population in the United States expands the problem facing parents, educators, and policymakers of how to manage children's out-of-school time. The 21stCentury Community Learning Center (21st CCLC) funding for OST programs remains the major source of funding if not the sole source of program funding. Total reliance on 21st CCLC funding is problematic for two reasons. First, programs will not be sustainable over the long-term with a single funding stream. Out-of-school time programs need multiple streams of funding to survive and thrive as community supports for children, youth, and families. Second, changes in the 21st CCLC funding program will spell disaster for dependent OST programs.

Recently, the Arkansas Department of Education began considering plans to allow school districts to divert 21st CCLC funds to sustain programs other than before and after school programs. This proposed change would jeopardize 120 out-of-school time programs in Arkansas serving 17,000 children and youth. The culprit is the way federal waivers for No Child Left

Behind Act of 2001 (NCLB, 2002) is presented to state departments of education. Departments of Education at the state level have the ability to choose the option for more flexible funding of existing 21st CCLC funds which would allow diversion of funds to individual school district programs of choice.

The risk is 21st CCLC funds would be diverted to athletics or other mainstream programs at the discretion of school districts. The burgeoning after-school movement could not sustain itself under the impact of lost federal funding through 21st CCLC. Threats to funding must be dealt with by building strong and sustainable community partnerships with a recognizable core of dedicated school and community volunteers and by acquiring and maintaining long term relationships with funders. Multiple streams of funding ensure the long term viability of OST programs (21st CCLC, 2011).

Research question one. The first research question, what are the elements of quality out-of-school time programs for school age youth as perceived by the participants? The consensus of the staff is the focus of the OST program is on student achievement, homework completion and enrichment activities. The elements of quality identified by the participants include: (1) targeted middle school students; (2) caring relationships between OST staff and students; (3) mentor relationships between students and volunteers; (4) improved student expectations in educational outcomes; (5) student achievement demonstrated by grades, test scores and homework; and (6) program sustainability and outcomes.

Academics were the driving force of the program based on the interviews, observational data and program documents. The OST program is designed to be a Monday through Thursday program operating three hours a day with a snack and transportation provided after the regular

school day. This is supported by information obtained from the Student/Parent Handbook.

Research question two. The second research question, what is the impact of quality as perceived by administrators, teachers, and students in out-of-school time programs? Teachers expressed a need for more one-on-one time with targeted students. The Site Coordinator stated, "Giving students feedback and that pat on the back or college and high school students helping with that test or homework makes a difference." (SC-0103) The consensus of the staff is the district cannot financially support the program. The staff was unable to identify any plan for sustainability. The teachers expressed in the interviews the parents were working as a reason for the limited amount of parental involvement in the program.

The school district has made an investment for over 11 years in this program and has the data to support the need to ensure the longevity of the program. Data were supported by student growth and achievement. The program had a strong buy-in from the administration and the parents. The majority of the students were mandated because of benchmark scores. There are other students that participated in the program for enrichment activities. One of the favorite activities noted by several teachers was Archery which is made possible by a strong sustainable community partnership.

The Site Coordinator addressed program sustainability in the following manner: We asked the district to pick up the transportation cost after providing the data that it was worth it. There is no way the district can sustain the whole program. I do think we have the support of the district and anyway they can, they will kick in what they can. However, it is unrealistic to think the district can. Our parents can't afford to pay; we don't have any industry here. My hope is to keep finding funding out there whether it is 21 Century or other sources. The transportation piece was big. We run three buses; intermediate, middle, and high school. We service the city whereas the other district services the county. As you look at the AYP scores, it has been a lot of hard work and a lot of knowledge. (SC-0103)

During the interview process the Site Coordinator expressed a major concern regarding the scope and depth of the program because of perceived funding restrictions and lack of community financial support.

Interpretation of Data

Patterns and relationships of data emerged. According to Miller (2003) out-of-school time programs play a vital role in identifying and addressing the needs of students and parents. However, the structure, personnel, program and funding are critical to high quality out-of-school time programs and must be more than an extension of the school day.

Therefore, a thorough analysis of data collected and further evaluation of documents, observation and interviews, the perspective of the participants are summarized in three key areas: First, the students' perceptions or understandings of the program were linked to completion of homework assignments, studying for tests, and participating in enrichment activities. Second, staff perceptions of program outcomes from the interviews were supported by student attendance, student participation, test scores and grades. Third, staff perceptions of program funding were supported by documentation of the funding cycle linked to program sustainability and the future of the out-of-school time program within the school district.

Positive gains were evident in the student's benchmark scores, yet the social and behavior outcomes could only be evaluated by observation of students and their level of engagement in activities of interest and involvement with peers and staff. The examination of student attendance records provided evidence of attendance and suggested a relationship between attendance and student achievement outcomes. Enrichment activities that build the self-esteem of students were difficult to evaluate based on the time period for program observation and the individual characteristics of the students.

Themes for each major category materialized during the process of analyzing interview data. The student category themes included the elements of safety, structure, and program. The student achievement category themes included grades, homework, and choices. The program sustainability and outcome category themes include elements of funding, support, and partners. For the relationship category, themes included the elements of caring and strength identified by the recurring key words strong and expectations.

The impact of quality was minimal on parental involvement in the out-of-school time program. Teachers reported parents were working which limited their involvement in the program. However, teachers felt that parents understand the benefits of the out-of-school program in relationship to home work help and test scores.

The review of literature did not identify a model or streamline formula for creating quality out-of-school time programs due to the various types and location. Elements of quality were identified in the target program of the study as described in the themes. The majority of the programs discussed in the review of literature had implemented some of the key quality elements and address the goal of creating student success through enrichment activities, academic achievement and personal development.

Program Recommendations

Program recommendations are four fold: (1) develop a sustainability plan to include employers, federal funds, community and faith-based organizations; (2) conduct student surveys to determine the perspective and interest of the students; (3) evaluate program models based on the needs of the students; and (4) empower the students to become advocates for the program through the use of technology and social media.

The sustainability plan includes the identification of employers, foundations or corporate offices that provide grant funds or volunteer hours/resources. The second part of the plan includes staff training on researching community, state and national funding sources and grant-writing techniques. Although, the faith-based organizations were involved, the partnership needs to be expanded to reach the sustainability goals. One African-American church in particular became the catalyst for this OST program because of their involvement in tutoring African-American students for the benchmark. More importantly, the OST program is a part of the school district's Arkansas Comprehensive School Improvement Plan (ACSIP) which is administered by the school district's federal program staff. The structure and allocation of federal funds may need to be realigned to include the OST program. The district reaps the benefits of the OST program as students become proficient with the assistance of the OST staff. As a result, the school district remains off the School Improvement List.

Out-of-school programs must address the needs of the whole child by using a program model that helps staff identify the needs of the students and establish a framework for the program and the student. The integration of technology into the program through service learning projects may provide meaningful and real-life experiences for the students. The students must also be empowered to accept responsibility and ownership for the program which may be a shift in roles for the staff.

Recommendations to the Field

For OST programs to become quality and sustainable within school districts and community organizations, (1) the role of OST programs may need redefining to address the needs of today's children and families, (2) community colleges and universities need to become more involved in shaping the role of out-of-school time programs, (3) the continuation of the advocacy work for legislation that support the need for out-of-school time programs and program expansion in highly populated areas of the state or areas with limited resources, and (4) the implementation of quality standards in out-of-school time programs must be a funded mandate regardless of the type of out-of-school time program or the location.

Employing quality elements for OST programs work on multiple levels. Building a strong foundation through multiple funding sources to sustain programming is essential to OST. Programmers should access private funding to build stronger programs. Creating cooperative community relationships bridge the gap between staff capability and program needs by bringing in volunteers. Volunteers can be identified through community partners or parents showing interest in program goals. Children and youth flourish in environments conducive to learning. The environment should feel safe to participants and be safe for participants.

The perception of safety by participants is as important as having planned evacuation routes and attendance tracking systems. Once participants feel safe they can begin to engage in programming and start building positive interpersonal relationships with staff and volunteers. The interactions between participants and staff are essentially where program goals and objectives are addressed. These interactions provide essential moments for mentoring, coaching, and teaching. The establishment of interpersonal relationships between participants and staff builds trust. Once trust is established, participants can reveal needs to staff and volunteers that can be addressed through the program. Trust also makes it easier to address developmental needs in a less threatening environment.

Having basic needs met frees participants to aspire to academic attainment and other personal goals. The perception of participants the program has developed specially to meet their needs allows participants to have ownership of the program. The perception of safety allows participants to build trust and form vital relationships which will meet needs. Meeting developmental needs leads to academic improvement which over time can spell success for the student both in and out of school.

Recommendations for Further Research

Ongoing studies are needed to examine the use of theory-based evaluations in out-ofschool time programs. Theory-based evaluation identifies the strengths and weaknesses of the program model and the evaluator targets the problem areas and program results or outcomes. Second, additional work is needed regarding the quality elements in OST program that can be replicated and show evaluative outcomes. The importance of evaluating program effectiveness is critical to sustainable out-of-school time programs, particularly 21st Century Community Learning Center Programs and other programs that are receiving federal funds. Third, additional research is needed to determine the number of hours and days that are most beneficial to program participants in out-of-school time programs. Last, one of the common threads in the review of literature is centered on the question of how to create and implement a system of quality standards, supports, and resources for sustainable out-of-school time programs that will be effective regardless of program type or location.

Conclusion

The No Child Left Behind Act of 2001 (NCLB, 2002) mandates school accountability in the classroom and the out-of-school time programs, particularly those that are federally funded. Historically, out-of-school time programs were started to provide a safe place for children while parents worked. Quality out-of-school time programs continue to play a critical role in closing the achievement gap specifically for economically disadvantaged youth by extending the school day and providing enrichment and academic support in a safe environment. However, the mark of success in out-of-school time programs continues to be student retention and academic achievement. Elements of quality for OST programs begin with multiple funding sources and parent and community partners. The highest goal of quality OST programs is educational attainment which is paramount to success in life. Improvement in student performance in benchmark examinations and individual educational achievement is the mark of OST program success.

Some OST programs in school districts have seen the value and benefit of these programs and others must recognize quality programs are needed to improve academic success and should consider the allocation of federal or other discretionary funding for out-of-school time programs. Out-of-school time programs help to keep children safe by reducing the temptation to engage in risk-taking behaviors, criminal activities, and the use of drugs, alcohol and tobacco. Out-ofschool time programs address the critical need for helping working parents and providing academic and enrichment activities to promote the social, emotional and physical development of children.

Student achievement was identified as the richest category with related themes of grades and homework. Interview data revealed shared themes regarding the program for the participants' category. The case study identified elements of quality that support student achievement outcomes for school based programs including positive program perceptions, sustainable funding, and building strong interpersonal relationships. The findings suggest key elements of quality were present and may contribute to positive outcomes for students. Program sustainability was a major concern for staff and the future of the OST program. This study contributes to the data needed to identify OST quality elements across program types and geographical locations.

The data from interviews and documents supported academic achievement and the participation of students in enrichment activities. Interview results from the staff indicated the importance of strong interpersonal relationships between staff and students. OST programs in a similar context can benefit from this finding because often students are reached by people they have come to know and trust. Teaching, mentoring, and coaching can be accomplished while students learn archery, journal writing, and mathematic remediation. Application of key elements to build a quality framework for OST programs are suggested within the interview data.

OST programs offer important partnerships with colleges and universities that provide struggling students with caring and supportive mentoring relationships. Strong programs emerge through hybrid programs utilizing a combination of faith-based and school-based programming efforts. This case study had a faith-based partnership at its point of origin that set the stage for school-based programming focused on increasing benchmark scores. This elemental focus on academic achievement suggested increased test scores. The case study shows the program may be stifled through the lack of vision toward expansion of faith-based and community partnerships that could increase sustainability and funding. Overall, without the 21st Century Community Learning Center funding, the site coordinator indicated the program would have to close its doors. The study identified key elements that build upon each other to create programming that effectively address student achievement.
References

- Afterschool Alliance. (2004). America after 3 PM: A household survey on afterschool programs in America. Working families and afterschool. A special report from America after 3 PM: Author. Retrieved from http://www.afterschoolalliance.org/press archives/america 3 pm/Executive Summary.pdf
- American Youth Policy Forum. (2006). *Helping youth succeed through out-of-school time programs*. Washington, DC: Author.
- Arkansas Department of Education. (2010). Arkansas public school profile. Retrieved from http://www.adearc.com/ade/Profile.asp?lea=2203000&year=2010
- Arkansas Department of Education. (2011). What every educators needs to know about common core state standards. Retrieved from http://arkansased.org/educators/pdf/curriculum/ccss_resource_062111.pdf-Windo
- Arkansas Out of School Network. (2009). *Arkansas standards for quality afterschool programs*. Little Rock, AR: Author.
- Arkansas Out of School Network. (2010). Afterschool is key! Little Rock, AR: Author.
- Arkansas School Performance Report. (2009). School report card 2009. Retrieved from http://normessasweb.uark.edu/schoolperformance/School/SRCy3.php
- Ascher, C. (2006). NCLB's supplemental educational services: Is this what our students need? *Phi Delta Kappan, 10*, 136-141.
- Baker, S., Speilberger, J., Lockaby, T., Guterman, K. (2010). Youth development: Enhancing quality in afterschool programs: Chicago, IL: Chapin Hall at the University of Chicago.
- Barth, J., & Nitta, K. (2008). Education in the post-lake view era: What is Arkansas doing to close the achievement gap? Little Rock, AR: University of Arkansas Clinton School of Public Service.
- Beebe, M. (2008). *Enriching Arkansas children's lives through high-quality out-of-school activities*. Little Rock, AR: Final report of the Governor's Task Force on Best Practices for After-school and Summer Programs.
- Bernard, R. H. (2000). *Social research methods: Qualitative and quantitative approaches*. Thousand Oak, CA: Sage Publications.

- Beverly, S. G. (2001). Material hardship in the United States: Evidence from the survey of income and program participation. *Social Research*, *9*, 143-151.3.
- Bodilly, S., & Beckett, M. (2005). Making out-of-school time matter: Evidence for an action agenda. Santa Monica, CA: Rand Corp.
- Brantlinger, E. A. (1997, April). *Knowledge, position, and agency: Activism and inward gaze as a natural next step in local inquiry.* Paper presented at the annual meeting of the American Educational Research Association, San Diego, CA.
- Carney, J., & Boyle, M. (2003). Stress, social support and depression in single and married mothers. *Social Psychiatry and Psychiatric Epidemiology*, *8*, 442-449.
- Cohen, M. A., & Piquero, A. R. (2007). New evidence on the monetary value of saving a high risk youth. Retrieved from www.towfoundation. org
- Children Now. (2001). After-school care for children: Challenges for California. Oakland, CA: Author.
- Community Matters & Breslin, T. (2003). Cost and scale in out-of-school time. Providence, RI: CM. Retrieved from mypasa.org/failid/Citywide_OST_Init.pdf
- Creswell, J. W. (2007). *Qualitative inquiry and research design: Choosing among five approaches*, (2nd ed.). Thousand Oaks, CA: Sage Publications.
- Creswell, J. W. (2009). *Research design: Qualitative, quantitative, and mixed method approaches,* (3rd ed.). Thousand Oaks, CA: Sage Publications.
- Cuff, S. P., Keown, R. E., Addy, C. L., & Garrison, C. G. (2005). Family psychological risk factors in a longitudinal epidemiology study of adolescents. *Journal of American Academic Child Adolescent Psychiatry*, 44, 121-129.
- Eccles, J., & Appleton, J. (2002). Community programs to promote youth development. Washington, DC: National Academy Press. Retrieved from http://www.nap.edu/catalog/10022.html
- Economic Development Commission. (2010). City economic development commission welcome. Retrieved from http://www.monticelloedc.org/
- Eisenhardt, K. M. (1989). Building theories from case study research. *The Academy of Management Review*, *14*(4), 532-550.
- Fashola, O. S. (2002). *Building effective afterschool program*. Thousand Oaks, CA: Corwin Press.

Firebaugh, G. (2008). Seven rules of social research. Princeton, NJ: Princeton University Press.

- Guzzardi, J., Little, M., & Mitchell, J. (2008). *The demand for after-school programs in Arkansas*. Little Rock, AR: University of Arkansas Clinton School of Public Service.
- Hall, G. (2002). After-school issues: *Citywide after-school initiative share their experiences developing and implementing standards*. Wellesley, MA: National Institute on Out-of School Time.
- Halpern, R. (2000). The promise of after-school programs for low-income children. *Early Childhood Research Quarterly*, 15(2), 185-214.
- Halpern, R. (2002). A different kind of child development institution: The history of afterschool programs for low-income children. *Teachers College Records*, 104(2), 178-211. Retrieved from http://www.tcrecord.org/pdf/10823.pdf
- Huang, D., Gribbons, B., Kim, K. S., Lee, C., Baker, E. L. (2000). A decade of results: The impact of the LA's best after-school enrichment program on subsequent student achievement performance. Los Angeles: CA: UCLA Center for the Study of Evaluation.
- Kane, T. J. (2004). *The impact of after-school programs: Interpreting the results of four recent evaluations*. Los Angeles, CA. University of California.
- Kugler, M. (2001). After-school programs are making a difference. *NASSP Bulletin*, 85(626), 3-21.
- Little, P., Wimer, C., Weiss, H., (2008). After-school programs in the 21st century: Issues and opportunities in out-of-school time evaluation. Retrieved from http://www.gse.harvard.edu/hfrp/projects/*afterschool*/resources/issuebriefs10.html
- Marshall, C. & Rossman, G. (2006). *Designing qualitative research*, (4th ed.). Thousand Oaks, CA: Sage Publications.
- Maslow, A. H. (1943). *The theory of human motivation*, (3rd ed.). Cambridge, NY: Harper & Row.
- Maslow, A. H. (1954). Motivation and personality, (3rd ed.). Cambridge, NY: Harper & Row.
- Merriman, S. B. (2009). *Qualitative Research: A Guide to Design and Implementation*, (3rd ed.). San Franciso, CA: Jossey-Bass.
- Miller, B. M. (2003). *Critical hours: Afterschool programs and educational success*. Brookline, MA: Miller Midzik Research Associates for the Nellie Mae Foundation. Retrieved from http://www.nmefdn.org/CriticalHours.htm

- Miller, K., Snow, D., & Lauer, P. (2004). *Out-of-school time programs for at-risk students*. Aurora, CO: Mid-continent Research for Education and Learning.
- National Center for Education Statistics. Common core of data public school data 2008-2009 school year: Host Middle School. Retrieved from http://nces.ed.gov/ccd/schoolsearch/school detail.asp?Search=1&Miles+10&Zip
- National Governors Association. (2006). Ten states held governors' summits on extra learning opportunities in 2005-06. Retrieved from http://www.nga.org/Files/pdf/0702ELOSUMMITS.PDF
- National School-Age Care Alliance. (1998). *The national school age care alliance standards for quality school-age care*. Boston, MA: Author. Retrieved from http://www.nsaca.org
- No Child Left Behind Act of 2001, Pub. L. No. 107-110 (2002).
- Office of Juvenile Justice and Delinquency Prevention. (2006). OJJDP statistical briefing book. Retrieved from http://ojjdp.ncjrs.gov/ojstatbb/offeders/qu03301asp
- Policy Studies Associates. (2005). Shared features of high-performing after-school programs: a follow-up to the TASC evaluation. Washington, DC: Author.
- Peter, N. (2002). *Outcomes and research in out-of-school time program design*. Philadelphia, PA: Best Practices Institute.
- Riley, R. W. (2000). Address: *Providing quality afterschool learning opportunities for America's families*. Washington, DC: United States Department of Education. Retrieved http://www.ed.gov/pubs/Providing Quality Afterschool Learning/report.html
- Scott-Little, C., Hamann, M. S., & Jurs, S. G. (2002). Evaluation of afterschool programs: A meta-evaluation of methodologies and narrative synthesis of finds. *American Journal of Evaluation*, 23(4), 387-419.
- Stake, R. (1995). The art of case study research. Thousand Oaks, CA: Sage Publications.
- Teachman, J. D. (2004). The childhood living arrangements of children and the characteristics of their marriages. *Journal of Family Issues*, 1, 86-111.
- Tolman, J., Pittman, K., Yohalem, N., Thomases, J., & Trammel, M. (2002). Moving an out-ofschool agenda, Task Brief #3: Quality standards, assessments and supports, Tacoma Park, MD: The Forum for Youth Investment. Retrieved from http://www.forumforyouthinvestment.org/grap/taskbrief3.pdf

- Trostil, P. A. (2007). The fiscal impact of college attainment. *New England Public Policy Center*. Retrieved from http://www.bos.frb.org/economic/neppc/wp/2007/neppcwp0702.html
- U.S. Census. (2009). American community survey 2008: l year estimate. Washington, DC: United States Government. Retrieved from http://census.gov/acs/www/data_documentation/2010_release/
- U.S. Census. (2000). American fact finder: Demographic profile highlights: Washington, DC: United States Government. Retrieved from http://factfinder.census.gov/servlet/SAFFFacts?_event=Searcj&geo_id&geoContext=&
- U.S. Census. (2000a). Households and families. Washington, DC: United States Government. Retrieved from http://factfinder.census.gov/servlet/GCTTable?_bm=y&geo_id=04000US05&-_box_head_
- U.S. Department of Education. (2002). Public law, print of PL 107-110, the no child left behind act (NCLB) of 2001. Washington, DC: Author. Retrieved from http://www.ed.gov/print/policy/elsec/leg/esea02/index.html
- U.S. Department of Education. (2002b). 21st Century community learning centers guidelines, Title IV, Part B, Section 4201, no child left behind act (NCLB) of 2001. Washington, DC: Author. Retrieved from http://www2.ed.gov/print/policy/elsec/leg/esea02/pg55.html
- U.S. Department of Education. (2003b). *When schools stay open late: The national evaluation* of the 21st century community learning centers program: First year findings. Washington, DC: Author. Retrieved from http://www.ed.gov/print/policy/elsec/leg/esea02/index.html
- U.S. Department of Health and Human Services Administration for Children and Families. (2010). Quick facts on school-age care: Program and practitioner standards. Washington, DC: Author. Retrieved from http://www.nccic.acf.hhs.gov/afterschool/program trends.html
- Vandell, D. L., Riesner, E. R., & Pierce, K. M. (2007). Outcomes linked to high-quality after-school programs: Longitudinal findings from the study of promising practices. Irvine, CA: University of California, and Washington, DC: Policy Studies Associates. Retrieved from http://www.gse.uci.edu/docs/PASP%20Final%20Report.pdf